# POLITECNICO DI TORINO Repository ISTITUZIONALE

A general review on the NLS equation with point-concentrated nonlinearity

Original A general review on the NLS equation with point-concentrated nonlinearity / Tentarelli, Lorenzo In: COMMUNICATIONS IN APPLIED AND INDUSTRIAL MATHEMATICS ISSN 2038-0909 14:1(2023), pp. 62-84. [10.2478/caim-2023-0004]
Availability: This version is available at: 11583/2983038 since: 2023-10-16T09:52:32Z
Publisher: Sciendo
Published DOI:10.2478/caim-2023-0004
Terms of use:
This article is made available under terms and conditions as specified in the corresponding bibliographic description in the repository
Publisher copyright

(Article begins on next page)





ISSN 2038-0909

DOI: 10.2478/caim-2023-0004

Review Article

# A general review on the NLS equation with point-concentrated nonlinearity

### Lorenzo Tentarelli<sup>1\*</sup>

<sup>1</sup>Politecnico di Torino, Dipartimento di Scienze Matematiche "G.L. Lagrange", Corso Duca degli Abruzzi 24, 10129, Torino, Italy

\*Email address for correspondence: lorenzo.tentarelli@polito.it

Communicated by Simone Dovetta Received on 04 18, 2023. Accepted on 07 28, 2023.

#### Abstract

The paper presents a complete (to the best of the author's knowledge) overview on the existing literature concerning the NLS equation with point-concentrated nonlinearity. Precisely, it mainly covers the following topics: definition of the model, weak and strong local well-posedness, global well-posedness, classification and stability (orbital and asymptotic) of the standing waves, blow-up analysis and derivation from the standard NLS equation with shrinking potentials. Also some related problem is mentioned.

Keywords: NLS equation, concentrated nonlinearity, delta potentials, well-posedness, standing waves, stability, blow-up, point-like limit

AMS subject classification: 35Q40, 35Q55, 35R06, 81Q99

#### 1. Introduction

The standard NonLinear Schrödinger (NLS) equation, i.e.,

(1) 
$$i\frac{\partial \psi}{\partial t} = -\Delta \psi + \beta |\psi|^{2\sigma} \psi, \qquad \beta \in \mathbb{R} \setminus \{0\}, \quad \sigma > 0, \quad \text{on } \mathbb{R}^+ \times \mathbb{R}^d,$$

is known to play a relevant role in several sectors of physics, where it appears as an effective evolution equation describing the evolution of a microscopic system on a macroscopic or mesoscopic scale. A typical example is provided by Bose-Einstein Condensates (BEC), whose behavior is "well approximated" in some suitable sense by the solutions of (1) (see, e.g., [1]). There are, however, other examples in which the physical meaning of the NLS equation is totally different, such as, for instance, the propagation of light in nonlinear optics, the behavior of water or plasma waves, the signal transmission through neurons (FitzHugh-Nagumo model), etc. (see [2] and references therein).

This paper is, to the best of our knowledge, a complete overview on the literature concerning one of the possible singular perturbations of (1): the NLS equation with *point-concentrated* nonlinearity (or Schrödinger equation with *nonlinear delta potential*). Such equation reads as (1), but with the nonlinearity formally multiplied times a *Dirac delta measure* based at the origin, i.e.

(2) 
$$i\frac{\partial \psi}{\partial t} = -\Delta \psi + \beta |\psi|^{2\sigma} \psi \delta_0, \qquad \beta \in \mathbb{R} \setminus \{0\}, \quad \sigma > 0, \quad \text{on } \mathbb{R}^+ \times \mathbb{R}^d.$$

The idea of introducing point perturbations of delta-type in Quantum Mechanics traces back to 1936 and is due to E. Fermi, though in a completely different context as he proposed linear point perturbations of delta-type of the Schrödinger equation (i.e.,  $\sigma = 0$  above) to model the interaction between a slow neutron and a fixed atom (see [3]).

The interest for the nonlinear model depicted in (2), on the contrary, is much more recent (and related to the applications of the standard NLS equation). It has been introduced in the 1990s to describe phenomena mostly related to solid state and condensed matter physics: charge accumulation in semiconductor interfaces or heterostructures ([4–9]), nonlinear propagation in a Kerr-type media in presence of

localized defects ( [10–12]), BEC in optical lattices where an isolated defect is generated by a focused laser beam ( [13,14]). Other applications are also suggested in acoustic, conventional and high- $T_c$  superconductivity, light propagation in photonic crystals, etc. (see, e.g., [11] and references therein). However, a rigorous mathematical derivation of the model in these contexts is still missing.

Again from the perspective of modeling BEC in presence of defects or impurities (whose spatial scale is supposed to be much smaller than the dispersion of the wave function) it is worth mentioning that also another equation involving delta potentials has been proposed in recent years ([15,16]):

(3) 
$$i\frac{\partial \psi}{\partial t} = (-\Delta + \alpha \delta_0)\psi + \beta |\psi|^{2\sigma}\psi, \qquad \alpha \in \mathbb{R}, \quad \beta \in \mathbb{R} \setminus \{0\}, \quad \sigma > 0, \quad \text{on } \mathbb{R}^+ \times \mathbb{R}^d.$$

Results on (3), which are not presented in this review, are discussed by [17–26] in dimension one and by [27–32] in dimensions two and three (while [33] addresses the circle, [34] addresses the half-line and [35–37] concern some first studies on a mixed model between (2) and (3)).

Finally, we mention that models involving delta potentials, as the one addressed by this paper, are rigorously defined only for d = 1, 2, 3 (more details on this point will be provided in Section 2). Hence, we will tacitly assume this restriction on the space dimension d of  $\mathbb{R}^d$  throughout the review.

**Remark 1.1.** In this paper we only focus on the case where a *single* nonlinear delta potential is based at the *origin*, i.e.  $\beta |\psi|^{2\sigma} \psi \delta_0$ . However, anything can be suitably adapted to the case of a delta based in other points of  $\mathbb{R}^d$ . Moreover, some results can be adapted to the case of finitely many deltas. We decided to limit ourselves to (2) for the sake of simplicity, but, nevertheless, we will mention the cases in which the results are known to extend to the more general frameworks.

#### 1.1. Organization of the paper

The paper is organized as follows.

- Section 2 addresses the rigorous definition of the model, starting from the linear case (Section 2.1) to the nonlinear case (Section 2.2).
- Section 3 addresses local well-posedness of the Cauchy problem associated with (2), both in the weak and in the strong sense.
- Section 4 addresses global well-posedness of the Cauchy problem associated with (2).
- Section 5 addresses the classification and the stability (*orbital* in Section 5.1, *asymptotic* in Section 5.2) of the standing waves of (2).
- Section 6 addresses blow-up analysis, from the first results (Section 6.1) to the the more refined ones (Section 6.3), also mentioning the question of pseudoconformal invariance (Section 6.2).
- Section 7 addressed the derivation of (2) from the standard NLS equation in dimension one (Section 7.1) and three (Section 7.2).
- Section 8 briefly mentions some connected problems.

#### 2. Definition of the model

In order to give a rigorous meaning to (2) it is necessary to start from the linear analogous ( $\sigma = 0$ ) and, then, go back to the nonlinear problem. According to what we said before, we limit ourselves to present the setting in the case of a single delta based at the origin.

# 2.1. The linear model

The most suitable way to give a rigorous meaning to a perturbation of delta type of the Laplacian, namely to an operator of the type  $-\Delta + \alpha \delta_0$ , with  $\alpha \in \mathbb{R}$ , is applying the theory of *self-adjoint extensions* of symmetric operators to  $-\Delta_{|C_0^{\infty}(\mathbb{R}^d\setminus\{0\})}$  (see, e.g., [38] for a general dissertation and [39] for some basics). In particular, one can prove that nontrivial extensions exist only for d = 1, 2, 3 and that, for any fixed

 $\alpha \in \mathbb{R}$ , the unique self-adjoint realization of  $-\Delta + \alpha \delta_0$  is given by the operator  $\mathcal{H}^d_{\alpha} : L^2(\mathbb{R}^d) \to L^2(\mathbb{R}^d)$  with domain

(4) 
$$\operatorname{dom}(\mathcal{H}_{\alpha}^{d}) := \left\{ u \in L^{2}(\mathbb{R}^{d}) : \exists \lambda > 0, \ \exists q \in \mathbb{C} \text{ such that} \right.$$

$$\left. u - \kappa_{d}(\alpha) q \mathcal{G}_{\lambda}^{d} =: \phi_{\lambda} \in H^{2}(\mathbb{R}^{d}) \text{ and } \phi_{\lambda}(0) = \theta_{\lambda}^{d}(\alpha) q \right\}$$

and action

(5) 
$$\mathcal{H}_{\alpha}^{d}u := -\Delta\phi_{\lambda} - \kappa_{d}(\alpha)q\lambda\mathcal{G}_{\lambda}^{d}, \qquad \forall u \in \text{dom}(\mathcal{H}_{\alpha}^{d}),$$

where

(6) 
$$\kappa_{d}(\alpha) := \begin{cases} -\alpha, & \text{if } d = 1, \\ 1, & \text{if } d = 2, 3, \end{cases} \qquad \theta_{\lambda}^{d}(\alpha) := \begin{cases} \frac{2\sqrt{\lambda}}{\alpha + 2\sqrt{\lambda}}, & \text{if } d = 1, \\ \alpha + \frac{\log\left(\frac{\sqrt{\lambda}}{2}\right) + \gamma}{2\pi}, & \text{if } d = 2, \\ \alpha + \frac{\sqrt{\lambda}}{4\pi}, & \text{if } d = 3, \end{cases}$$

with  $\gamma$  the Euler-Mascheroni constant, and  $\mathcal{G}^d_{\lambda}$  is the Green's function of  $-\Delta + \lambda$ , i.e.

(7) 
$$\mathcal{G}_{\lambda}^{d}(x) = \mathscr{F}^{-1}\left(\frac{1}{(2\pi)^{d/2}(|k|^{2} + \lambda)}\right)(x) = \begin{cases} \frac{e^{-\sqrt{\lambda}|x|}}{2\sqrt{\lambda}}, & \text{if } d = 1, \\ \frac{K_{0}(\sqrt{\lambda}|x|)}{2\pi}, & \text{if } d = 2, \\ \frac{e^{-\sqrt{\lambda}|x|}}{4\pi|x|}, & \text{if } d = 3, \end{cases}$$

with  $\mathscr{F}$  denoting the unitary Fourier transform of  $\mathbb{R}^d$  (i.e.,  $\mathscr{F}(h)[k] := (2\pi)^{-d/2} \int_{\mathbb{R}^d} \mathrm{e}^{-\imath k \cdot x} h(x) \, dx$ , whenever  $h \in L^1(\mathbb{R}^d)$ ) and  $K_0$  denoting the modified Bessel function of the second kind of order 0, also known as  $Macdonald\ function$  (see, e.g., [40, Section 9.6]). Note that  $\mathrm{dom}(\mathcal{H}^d_\alpha)$  is a Hilbert space if endowed with the  $graph\ norm\ \|\cdot\|^2_{\mathrm{dom}(\mathcal{H}^d_\alpha)} := \|\cdot\|^2_{L^2(\mathbb{R}^d)} + \|\mathcal{H}^d_\alpha\cdot\|^2_{L^2(\mathbb{R}^d)}$ .

**Remark 2.1.** Note that, when d=1,  $\alpha=0$  corresponds to the free Laplacian. On the contrary, it can be recovered in d=2,3 only letting  $\alpha\to\infty$ .

As a consequence of (4), any function in  $\operatorname{dom}(\mathcal{H}_{\alpha}^d)$  admits a decomposition in a regular part  $\phi_{\lambda}$ , on which the operator acts as the standard Laplacian, and a singular part  $q\mathcal{G}_{\lambda}^d$ , on which the operator acts as the multiplication times  $-\lambda$ . In addition, we recall that the two components are bound by the so-called boundary condition  $\phi_{\lambda}(0) = \theta_{\lambda}^d(\alpha) q$  and that the strength q of the singular part is usually called charge. The charge is uniquely determined for any  $u \in \operatorname{dom}(\mathcal{H}_{\alpha}^d)$  and different charges identify different functions  $u \in \operatorname{dom}(\mathcal{H}_{\alpha}^d)$  (see, e.g., [27, Remark 2.1] for the case d = 2). On the contrary,  $\lambda$  is a dumb parameter in the sense that every function of  $\operatorname{dom}(\mathcal{H}_{\alpha}^d)$  admits an equivalent decomposition for any positive value of  $\lambda$ , due to the fact that the difference between Green's functions with different values of  $\lambda$  is in  $H^2(\mathbb{R}^d)$  (see again [27, Remark 2.1]).

As a further evidence that the parameter  $\lambda$  cannot actually affect the definition of  $\mathcal{H}_{\alpha}^d$ , it is possible to find formulations equivalent to (4) and (5) that do not involve  $\lambda$ . In other words, one can check that

 $dom(\mathcal{H}_{\alpha}^d)$  is equal to

(8) 
$$\left\{ u \in H^1(\mathbb{R}) \cap H^2(\mathbb{R} \setminus \{0\}) : u'(0^+) - u'(0^-) = \alpha u(0) \right\}, \quad \text{if} \quad d = 1,$$

(9) 
$$\left\{ u \in L^2(\mathbb{R}^2) : \exists q \in \mathbb{C} \text{ s.t. } u - q\mathcal{G}_0^2 =: \phi \in H^2_{loc}(\mathbb{R}^2) \cap \mathring{H}^2(\mathbb{R}^2), \ \phi(0) = \alpha q \right\}, \quad \text{if} \quad d = 2,$$

$$(10) \quad \left\{ u \in L^{2}(\mathbb{R}^{3}) : \exists q \in \mathbb{C} \text{ s.t. } u - q\mathcal{G}_{0}^{3} =: \phi \in H_{loc}^{2}(\mathbb{R}^{3}) \cap \mathring{H}^{1}(\mathbb{R}^{3}) \cap \mathring{H}^{2}(\mathbb{R}^{3}), \ \phi(0) = \alpha q \right\}, \quad \text{if} \quad d = 3,$$

where

(11) 
$$\mathcal{G}_0^d(x) = \mathcal{F}^{-1}\left(\frac{1}{(2\pi)^{d/2}|k|^2}\right)(x) = \begin{cases} -\frac{\log|x|}{2\pi}, & \text{if } d=2, \\ \frac{1}{4\pi|x|}, & \text{if } d=3, \end{cases}$$

and that

(12) 
$$\mathcal{H}_{\alpha}^{d}u = \begin{cases} -\frac{d^{2}u}{dx^{2}}, & \text{in } \mathbb{R}\backslash\{0\}, & \text{if } d = 1, \\ -\Delta\phi, & \text{in } \mathbb{R}^{d}\backslash\{0\}, & \text{if } d = 2, 3, \end{cases} \quad \forall u \in \text{dom}(\mathcal{H}_{\alpha}^{d}).$$

In view of the previous definitions, it is not evident that  $\mathcal{H}^d_{\alpha}$  is the proper (self-adjoint) realization of the delta perturbation of the Laplacian. However, easy computations yields that, in the sense of distributions,

$$\mathcal{H}_{\alpha}^{d}u = -\Delta u - \kappa_{d}(\alpha)q\delta_{0}, \quad \forall u \in \text{dom}(\mathcal{H}_{\alpha}^{d}).$$

Note also that, when d=1, there results q=u(0), so that one can rigorously claim that  $\mathcal{H}_{\alpha}^{d}=-\frac{d^{2}}{dx^{2}}+\alpha\delta_{0}$ . Thus, the construction provided before is not strictly necessary for d=1. This is due to the fact that  $-\frac{d^{2}}{dx^{2}}+\alpha\delta_{0}$  is a form bounded perturbation of the Laplacian, which makes the one-dimensional case considerably different from the other ones because of the KLMN Theorem (see, e.g., [41, Theorem X.17]).

Remark 2.2. It is worth mentioning that the one-dimensional case is special also from another point of view. Indeed, only in this case, the one described above is not the unique class of nontrivial self-adjoint extension of  $-\Delta_{|C_0^{\infty}(\mathbb{R}\setminus\{0\})}$ . In fact, when d=1 there is a four parameters family of self-adjoint extensions (while in d=2, 3 there is only a one parameter family of self-adjoint extensions). However, these further generalizations are not of delta type and go beyond the aims of the present review, so that they are omitted.

Remark 2.3. Another way to obtain delta perturbations of the Laplacian is by studying the asymptotic behavior of perturbations of the Laplacian consisting of smooth suitably shrinking potentials (both local and nonlocal). This can be found, for instance, in [38] and brings exactly to the operator  $\mathcal{H}^d_{\alpha}$  introduced before.

The spectrum  $\sigma(\mathcal{H}_{\alpha}^{d})$  varies with the dimension as follows

(13) 
$$\sigma(\mathcal{H}_{\alpha}^{d}) := \begin{cases} \left\{ [0, +\infty), & \text{if } \alpha \geq 0, \\ \{\ell_{\alpha}^{1}\} \cup [0, +\infty), & \text{if } \alpha < 0, \end{cases} \right\} & \text{if } d = 1, \\ \left\{ \ell_{\alpha}^{2}\} \cup [0, +\infty), & \text{if } d = 2, \\ \left\{ [0, +\infty), & \text{if } \alpha \geq 0, \\ \{\ell_{\alpha}^{3}\} \cup [0, +\infty), & \text{if } \alpha < 0, \end{cases} & \text{if } d = 3, \end{cases}$$

with

(14) 
$$\ell_{\alpha}^{d} := \begin{cases} -\alpha^{2}/4, & \text{if } d = 1, \text{ and } \alpha < 0, \\ -4e^{-4\pi\alpha - 2\gamma} & \text{if } d = 2, \text{ and } \alpha \in \mathbb{R}, \\ -16\pi^{2}\alpha^{2}, & \text{if } d = 3, \text{ and } \alpha < 0. \end{cases}$$

where, as usual in the context of the operator theory, the presence or the absence of the negative eigenvalue  $\ell^d_{\alpha}$  distinguish between an attractive delta potential and a repulsive delta potential (note that in d=2 the delta potential is always attractive). In the cases in which  $\ell^d_{\alpha}$  is present its eigenspace is spanned by  $\mathcal{G}^d_{-\ell^d}$ .

Moreover, the quadratic form associated with  $\mathcal{H}_{\alpha}^{d}$  reads

(15) 
$$Q_{\alpha}^{d}(u) := \begin{cases} \left\| \frac{du}{dx} \right\|_{L^{2}(\mathbb{R})}^{2} + \alpha |q|^{2}, & \text{if } d = 1 \\ \|\nabla \phi_{\lambda}\|_{L^{2}(\mathbb{R}^{d})}^{2} + \lambda (\|\phi_{\lambda}\|_{L^{2}(\mathbb{R}^{d})}^{2} - \|u\|_{L^{2}(\mathbb{R}^{d})}^{2}) + \theta_{\lambda}^{d}(\alpha) |q|^{2}, & \text{if } d = 2, 3, \end{cases}$$

for every  $u \in \text{dom}(\mathcal{Q}_{\alpha}^d)$ , where

$$(16) \quad \operatorname{dom}(\mathcal{Q}_{\alpha}^{d}) = V_{d} := \begin{cases} H^{1}(\mathbb{R}), & \text{if} \quad d = 1, \\ \{u \in L^{2}(\mathbb{R}^{d}) : \exists \lambda > 0, \ \exists q \in \mathbb{C} \text{ s.t. } u - q\mathcal{G}_{\lambda}^{d} =: \phi_{\lambda} \in H^{1}(\mathbb{R}^{d})\}, & \text{if} \quad d = 2, 3. \end{cases}$$

We introduced the notation  $V_d$  for the form domain to underline that, in contrast to the operator domain, it does not depend on  $\alpha$ . In addition,  $V_d \supseteq H^1(\mathbb{R}^d)$ , when  $d \neq 1$ , and it is a Hilbert space when endowed with the norm

$$||u||_{V_d}^2 := \begin{cases} ||u||_{H^1(\mathbb{R})}^2, & \text{if} \quad d = 1, \\ ||\phi_{\lambda}||_{H^1(\mathbb{R}^d)}^2 + |q|^2 ||\mathcal{G}_{\lambda}^d||_{L^2(\mathbb{R}^d)}^2, & \text{if} \quad d = 2, 3 \end{cases}$$

(equivalent for every fixed  $\lambda > 0$ , when d = 2, 3). In fact, in the three-dimensional case the quadratic form admits a representation analogous to that of the one-dimensional case, which arises letting  $\lambda \to 0$ , i.e.

$$\mathcal{Q}^3_{\alpha}(u) = \|\nabla \phi\|^2_{L^2(\mathbb{R}^3)} + \alpha |q|^2, \qquad \forall u \in V_3,$$

with

$$V_3 = \{ u \in L^2(\mathbb{R}^3) : \exists q \in \mathbb{C} \text{ s.t. } u - q\mathcal{G}_0^3 =: \phi \in H^1_{loc}(\mathbb{R}^3) \cap \mathring{H}^1(\mathbb{R}^3) \}.$$

In this case the norm can be equivalently written as  $||u||_{V_3}^2 = ||\nabla \phi||_{L^2(\mathbb{R}^3)}^2 + |q|^2$ . On the contrary, in the two-dimensional case it is not known whether this kind of representation is available due to the low regularity of the regular part when  $\lambda = 0$ . Such a difference is a consequence of the fact that  $\mathcal{G}^3_{\lambda} \to \mathcal{G}^3_0$ , while  $\mathcal{G}^2_{\lambda} \to +\infty$ , pointwise as  $\lambda \to 0$ .

**Remark 2.4.** We also mention that, whenever  $\ell^d_{\alpha}$  exists, it satisfies

$$\ell_{\alpha}^{d} = \mathcal{Q}_{\alpha}^{d} \left( f_{d}(-\ell_{\alpha}^{d}) \mathcal{G}_{-\ell_{\alpha}^{d}}^{d} \right) = \inf_{\substack{u \in V_{d} \\ \|u\|_{L^{2}(\mathbb{R}^{d})} = 1}} \mathcal{Q}_{\alpha}^{d}(u),$$

with  $f_d: \mathbb{R}^+ \to \mathbb{R}^+$  given by

$$f_d(\lambda) := \begin{cases} 2\sqrt[4]{\lambda^3}, & \text{if } d = 1, \\ 2\sqrt{\pi\lambda} & \text{if } d = 2, \\ 2\sqrt[4]{4\pi^2\lambda}, & \text{if } d = 3, . \end{cases}$$

Finally, we recall that, since  $\mathcal{H}^d_{\alpha}$  is self-adjoint, as a consequence of Stone's theorem, for every  $\psi_0 \in \text{dom}(\mathcal{H}^d_{\alpha})$ , there exists a unique function  $\psi \in C^0_{loc}([0,+\infty);\text{dom}(\mathcal{H}^d_{\alpha})) \cap C^1_{loc}([0,+\infty);L^2(\mathbb{R}^d))$  which strongly solves

(17) 
$$\begin{cases} i \frac{\partial \psi}{\partial t} = \mathcal{H}_{\alpha}^{d} \psi \\ \psi(0, \cdot) = \psi_{0} \end{cases}$$

in the sense that the former is satisfied as an equality in  $C^0_{loc}([0, +\infty); L^2(\mathbb{R}^d))$  and the latter as an equality in dom $(\mathcal{H}^d_{\alpha})$ . On the other hand, as the first equation in (17) is invariant under gauge transformations and time translations, the associated mass, i.e.

(18) 
$$M(t) = M(\psi(t, \cdot)) := \|\psi(t, \cdot)\|_{L^{2}(\mathbb{R}^{d})}^{2},$$

and energy, i.e.  $\frac{1}{2}\mathcal{Q}_{\alpha}^{d}(\psi(t,\cdot))$  in the linear case, are preserved along the flow. Moreover, there exists a  $L^{2}(\mathbb{R}^{d})$ -strongly continuous unitary group  $(U_{d}(t))_{t}$  such that  $\psi(t,\cdot) = U_{d}(t)\psi_{0}$ . However, since the explicit form of the kernel of  $U_{d}(t)$  is not easily manageable (see [42]), it is customary to represent the solution as

(19) 
$$\psi(t,x) = (\mathcal{U}_d(t)\psi_0)(x) + i \int_0^t \mathcal{U}_d(t-s,x)\kappa_d(\alpha)q(s) \, ds,$$

where  $\mathcal{U}_d(t)$  is the free propagator of  $\mathbb{R}^d$ , with kernel

(20) 
$$\mathcal{U}_d(t,x) := \mathscr{F}^{-1}(e^{-\imath |k|^2 t})[x] = \frac{e^{-\frac{|x|^2}{4\imath t}}}{(4\imath \pi t)^{d/2}},$$

and  $q(\cdot)$  is the unique solution of

(21) 
$$q(t) + \int_0^t \mathcal{K}_d(t-s)c_d(\alpha)q(s) ds = m_d f_d(t),$$

with

(22) 
$$c_{d}(\alpha) := \begin{cases} \frac{e^{i\pi/4}}{2\sqrt{\pi}}\alpha, & \text{if } d = 1, \\ 4\pi(\theta_{1}^{2}(\alpha) - i/8), & \text{if } d = 2, \\ 4\sqrt{\pi}e^{i\pi/4}\alpha, & \text{if } d = 3, \end{cases} \qquad m_{d} := \begin{cases} 1, & \text{if } d = 1, \\ 4\pi, & \text{if } d = 2, \\ c_{d}(1), & \text{if } d = 3, \end{cases}$$

(23) 
$$f_d(t) := \begin{cases} (\mathcal{U}_1(t)\psi_0)(0), & \text{if } d = 1\\ \int_0^t \mathcal{K}_d(t-s)(\mathcal{U}_d(s)\psi_0)(0) \, ds, & \text{if } d = 2, 3, \end{cases}$$

and

(24) 
$$\mathcal{K}_d(t) := \begin{cases} \frac{1}{\sqrt{t}}, & \text{if } d = 1, 3, \\ \int_0^{+\infty} \frac{t^{s-1}}{\Gamma(s)} ds, & \text{if } d = 2. \end{cases}$$

Since at any time t the solution q(t) of (21) is the coefficient of the singular part of  $\psi(t,\cdot)$ , with a little abuse of notation, it is called *charge* and thus (21) is called *linear charge equation*.

#### 2.2. From linear to nonlinear

In view of the previous section, the nonlinear problem arises simply letting the "strength"  $\alpha$  of the delta perturbation depend on the unknown  $\psi$ , and more precisely on the charge q, as follows:

(25) 
$$\alpha = \alpha(q) := \beta |q|^{2\sigma} \qquad \beta \in \mathbb{R} \setminus \{0\}, \quad \sigma > 0.$$

As a consequence, the associated Cauchy problem reads

(26) 
$$\begin{cases} i \frac{\partial \psi}{\partial t} = \mathcal{H}_{\beta,\sigma}^d \psi \\ \psi(0,\cdot) = \psi_0 \end{cases}$$

where  $\mathcal{H}_{\beta,\sigma}^d$  is now a nonlinear map with domain

(27) 
$$\operatorname{dom}(\mathcal{H}_{\beta,\sigma}^d) := \left\{ u \in L^2(\mathbb{R}^d) : \exists \lambda > 0, \exists q \in \mathbb{C} \text{ such that } \right\}$$

$$0, \exists q \in \mathbb{C} \text{ such that}$$

$$u - \kappa_d(\beta |q|^{2\sigma}) q \mathcal{G}_{\lambda}^d =: \phi_{\lambda} \in H^2(\mathbb{R}^d) \text{ and } \phi_{\lambda}(0) = \theta_{\lambda}^d \left(\beta |q|^{2\sigma}\right) q$$

and action

(28) 
$$\mathcal{H}_{\beta,\sigma}^d u = -\Delta \phi_{\lambda} - \kappa_d(\beta |q|^{2\sigma}) q \lambda \mathcal{G}_{\lambda}^d, \qquad \forall u \in \text{dom}(\mathcal{H}_{\beta,\sigma}^d),$$

with  $\kappa_d$  still defined by (6) (an equivalent definition can be constructed from (8)-(9)-(10)-(12)). Consistently with the linear case, strong solutions on an interval [0,T] of (27) have to be meant as functions in  $C^0([0,T]; \operatorname{dom}(\mathcal{H}^d_{\beta,\sigma})) \cap C^1([0,T]; L^2(\mathbb{R}^d))$  which satisfy the former as an equality in  $C^0([0,T]; L^2(\mathbb{R}^d))$  and the latter as an equality in  $\operatorname{dom}(\mathcal{H}^d_{\beta,\sigma})$ . Note that  $\operatorname{dom}(\mathcal{H}^d_{\beta,\sigma})$  is no more a linear space, but nevertheless a notion of convergence is still defined there by a natural analogue of the graph norm.

Furthermore, it is worth mentioning that, since in this context no general theory is available, the study of (26) relies on the study of the nonlinear versions of (19) and (21), that is

(29) 
$$\psi(t,x) = (\mathcal{U}_d(t)\psi_0)(x) + i \int_0^t \mathcal{U}_d(t-s,x)\kappa_d(\beta|q(s)|^{2\sigma})q(s) ds,$$

and

(30) 
$$q(t) + \int_0^t \mathcal{K}_d(t-s)c_d(\beta|q(s)|^{2\sigma})q(s) \, ds = m_d f_d(t).$$

More precisely, all the methods are based on proving that (30), also known as nonlinear charge equation, has a unique solution with sufficiently "nice" features so that the function  $\psi$  defined by (29) is a solution of (26) with the required regularity.

#### 3. Local well-posedness and conservation laws

In this section we present results concerning the local well-posedness of (26) and the associated conservation laws. The first results on local well-posedness of (26) that appeared in the literature concerned existence and uniqueness of weak solutions, where in this context a weak solution of (26) on an interval [0,T] is a function  $\psi$  such that

(31) 
$$\psi(t,\cdot) = \phi_{\lambda}(t,\cdot) + q(t)\mathcal{G}_{\lambda}^{d} \in V_{d}, \quad \forall t \in [0,T], \quad \text{and}$$

(31) 
$$\psi(t,\cdot) = \phi_{\lambda}(t,\cdot) + q(t)\mathcal{G}_{\lambda}^{\alpha} \in V_{d}, \quad \forall t \in [0,T], \quad \text{and}$$

$$\begin{cases}
i\frac{d}{dt}\langle \chi, \psi(t,\cdot)\rangle_{L^{2}(\mathbb{R}^{d})} = \langle \nabla \chi_{\lambda}, \nabla \phi_{\lambda}(t,\cdot)\rangle_{L^{2}(\mathbb{R}^{d})} + \lambda(\langle \chi_{\lambda}, \phi_{\lambda}(t,\cdot)\rangle_{L^{2}(\mathbb{R}^{d})} - \langle \chi, \psi(t,\cdot)\rangle_{L^{2}(\mathbb{R}^{d})}) \\
+ \theta_{\lambda}^{d} \left(\beta|q(t)|^{2\sigma}\right) q_{\chi}^{*}q(t), \quad \forall \chi = \chi_{\lambda} + q_{\chi}\mathcal{G}_{\lambda}^{d} \in V_{d}, \quad \forall t \in [0,T], \\
\psi(0,\cdot) = \psi_{0}, \quad \text{in} \quad V_{d}.
\end{cases}$$

Given this definition one can state the first well-posedness result.

**Theorem 3.1** (d = 1 in [43], d = 2 in [44], d = 3 in [45]). Let  $\beta \in \mathbb{R} \setminus \{0\}$ ,  $\sigma > 0$  and  $\psi_0 \in V_d$ . Therefore:

- (i) if d = 1, 3, then there exists T > 0 such that (26) admits a unique weak solution on [0, T];
- (ii) if d = 2,  $\sigma \ge 1/2$  and  $(1 + |k|^{\varepsilon})\mathscr{F}(\psi_0) \in L^1(\mathbb{R}^2)$ , for some  $\varepsilon > 0$ , then there exists T > 0 such that (26) admits a unique weak solution on [0,T]

In addition, letting

(33) 
$$T_{\beta,\sigma}^d(\psi_0) := \sup\{T > 0 : \text{there exists a unique weak solution of (26) on } [0,T]\},$$

there results that

(34) 
$$M(t) = M(0) \quad and \quad E_{\beta,\sigma}^d(t) = E_{\beta,\sigma}^d(0), \quad \forall t \in [0, T_{\beta,\sigma}^d(\psi_0)),$$

where M(t) is the mass associated with  $\psi(t,\cdot)$ , defined by (18), and  $E^d_{\beta,\sigma}(t)$  is the energy associated with  $\psi(t,\cdot)$ , defined by

(35) 
$$E_{\beta,\sigma}^d(t) = E_{\beta,\sigma}^d(\psi(t,\cdot)) :=$$

$$\begin{cases} \frac{1}{2} \|\hat{\partial}_x \psi(t,\cdot)\|_{L^2(\mathbb{R})}^2 + \frac{\beta |q(t)|^{2\sigma+2}}{2\sigma+2}, & \text{if } d=1 \\ \\ \frac{1}{2} \|\nabla \phi_{\lambda}(t,\cdot)\|_{L^2(\mathbb{R}^d)}^2 + \frac{\lambda}{2} (\|\phi_{\lambda}(t,\cdot)\|_{L^2(\mathbb{R}^d)}^2 - \|\psi(t,\cdot)\|_{L^2(\mathbb{R}^d)}^2) + \frac{\theta_{\lambda}^d \left(\frac{\beta |q(t)|^{2\sigma}}{\sigma+1}\right) |q(t)|^2}{2}, & \text{if } d=2, 3. \end{cases}$$

**Remark 3.1.** Conservations of mass and energy are not surprising since the invariances pointed out in the linear case are still present in the nonlinear case. Moreover, one can easily see that, setting  $\beta = \alpha$  and  $\sigma = 0$ , there results  $E_{\beta,\sigma}^d(t) = \frac{1}{2} \mathcal{Q}_{\alpha}^d(\psi(t,\cdot))$ . As a consequence, in d=3 it is possible to find a form of the energy where the parameter  $\lambda$  does not appear.

**Remark 3.2.** In fact, as a byproduct of the proof of Theorem 3.1, one has that the weak solution of (26) belongs to  $C^0([0,T];V_d)$ , for any  $T \in (0,T_{\beta,\sigma}^d(\psi_0))$ .

The further assumptions required in the two-dimensional case are mainly due to the different qualitative behavior of the integral kernel of the charge equation  $\mathcal{K}_d$ , defined by (24). Indeed, whereas in odd dimensions the  $\frac{1}{2}$ -Abel kernel possesses nice regularizing properties in Sobolev spaces, in the two-dimensional case the Volterra function of order -1 has no regularizing properties in those spaces due to its highly singular behavior at the origin (for details see [44,46]). Moreover, while in odd dimension Theorem 3.1 can be extended to finitely many nonlinear delta potentials, in the two-dimensional case the possibility of such an extension is open (and far from being understood).

Finally, again in odd dimension, it is possible to strengthen the results of Theorem 3.1. Note that the following statement does not coincide with the ones contained in the original sources. It is written in such a way to be consistent with the notation used in Theorem 3.1.

Theorem 3.2 (d = 1 in [47, case s = 1], d = 3 in [48]). Let  $d = 1, 3, \beta \in \mathbb{R} \setminus \{0\}$ ,  $\sigma > 0$  and  $\psi_0 \in \text{dom}(\mathcal{H}^d_{\beta,\sigma})$ . Then, there exists a unique strong solution of (26) on [0,T], for every  $T \in (0,T^d_{\beta,\sigma}(\psi_0))$  (with  $T^d_{\beta,\sigma}(\psi_0)$  defined by (33)).

**Remark 3.3.** In d = 2 such a result on strong solutions is still missing due again to the features of the kernel of the charge equation, mentioned above.

# 4. Global well-posedness

The study of global well-posedness consists of detecting those cases in which the quantity  $T^d_{\beta,\sigma}(\psi_0)$ defined by (33) is equal to  $+\infty$ ; namely, the cases in which the solution is global-in-time.

In this context, the crucial points usually are:

(i) proving the so-called blow-up alternative, i.e.

(36) 
$$T^d_{\beta,\sigma}(\psi_0) < +\infty$$
 and  $\limsup_{t \nearrow T^d_{\beta,\sigma}(\psi_0)} \|\psi(t,\cdot)\|_{V_d} < +\infty \implies \text{contradiction}$ 

or equivalently

(37) 
$$T_{\beta,\sigma}^d(\psi_0) < +\infty \quad \text{and} \quad \limsup_{t \nearrow T_{\beta,\sigma}^d(\psi_0)} |q(t)| < +\infty \quad \Longrightarrow \quad \text{contradiction};$$

(ii) proving that conservations of mass and energy entail a-priori boundedness for  $\|\psi(t,\cdot)\|_{V_d}$ , or |q(t)|, on  $[0, T_{\beta,\sigma}^d(\psi_0))$ .

For the sake of clarity we distinguish the result on the defocusing case, i.e.  $\beta > 0$ , from those on the focusing case, i.e.  $\beta < 0$ .

Theorem 4.1 (d = 1 in [43], d = 2 in [44], d = 3 in [45]). Let  $\beta$ ,  $\sigma > 0$  and  $\psi_0 \in V_d$  and let  $T_{\beta,\sigma}^d(\psi_0)$ be defined as in (33). Therefore:

- (i) if d=1, 3, then  $T^d_{\beta,\sigma}(\psi_0)=+\infty$ ; (ii) if  $d=2, \ \sigma \geqslant 1/2$  and  $(1+|k|^{\varepsilon})\mathscr{F}(\psi_0) \in L^1(\mathbb{R}^2)$ , for some  $\varepsilon > 0$ , then  $T^2_{\beta,\sigma}(\psi_0)=+\infty$ .

Moreover, in these cases the weak solution of (26) belongs to  $L^{\infty}([0,+\infty);V_d)$ .

**Remark 4.1.** Note that the presence of further assumptions when d=2 in Theorem 4.1 is only due to the fact that they are necessary in Theorem 3.1.

**Theorem 4.2** (d = 1 in [43], d = 3 in [45]). Let  $d = 1, 3, \beta < 0, \sigma \in (0, 1]$  and  $\psi_0 \in V_d$  and let  $T_{\beta,\sigma}^d(\psi_0)$  be defined as in (33). Therefore:

- (i) if  $\sigma < 1$ , then  $T_{\beta,\sigma}^d(\psi_0) = +\infty$ ; (ii) if  $\sigma = 1$ , then there exists  $\mu_{\beta}^d > 0$  such that  $T_{\beta,1}^d(\psi_0) = +\infty$  whenever  $M(\psi_0) < \mu_{\beta}^d$ .

Moreover, in these cases the weak solution of (26) belongs to  $L^{\infty}([0,+\infty);V_d)$ .

The limitations on  $\sigma$  displayed by Theorem 4.2 are sharp, in the sense that, as we will see in Section 6, beyond the threshold  $\sigma = 1$ , solutions which are not global-in-time may arise. As a consequence, the power  $\sigma = 1$  is usually called  $L^2$ -critical power, while smaller powers are called  $L^2$ -subcritical powers and larger powers are called  $L^2$ -supercritical powers. In addition, also the threshold on the mass  $\mu_{\beta}^d$ , present in the case  $\sigma = 1$ , is sharp in the same sense as before. As a consequence, such parameter is called  $L^2$ -critical mass. It can be in fact explicitly computed and reads:

(38) 
$$\mu_{\beta}^{d} := \begin{cases} -\frac{2}{\beta}, & \text{if } d = 1, \\ \frac{1}{-32\pi^{2}\beta}, & \text{if } d = 3. \end{cases}$$

Finally, it is worth mentioning that also the absence of d=2 in Theorem 4.2 is sharp. Indeed, as we will see again in Section 6, in the two-dimensional case solutions which are not global-in-time may be found for any value of  $\sigma$ . In other words, the reason for which there is no  $L^2$ -critical power in d=2 is that in this case any power is  $L^2$ -supercritical. Consistently, the concept of  $L^2$ -critical mass is not defined in d=2.

**Remark 4.2.** Concerning the extension of the previous theorems to the case of finitely many nonlinear deltas, this is known for Theorem 4.1(i), for Theorem 4.2(i) and, in the case d = 1, for Theorem 4.2(i).

#### 5. Standing waves

A further point of interest in the study of (26) is given by the *standing waves*, i.e. solutions of the equation of the form  $\psi_{\beta,\sigma,\omega}^d(t,x) = e^{i\omega t}u_{\beta,\sigma,\omega}^d(x)$ , for some  $\omega \in \mathbb{R}$  usually called *frequency*. Clearly, the search for such solutions reduces to the search for the functions

$$u_{\beta,\sigma,\omega}^d \in \text{dom}(\mathcal{H}_{\beta,\sigma}^d)$$
 that satisfy  $(\mathcal{H}_{\beta,\sigma}^d + \omega)u_{\beta,\sigma,\omega}^d = 0.$ 

These are usually called *bound states* of the NLS with concentrated nonlinearity, can be completely classified in any dimension and are positive up to the gauge invariance  $e^{i\theta}$ ,  $\theta \in [0, 2\pi)$ .

Theorem 5.1 (d = 1 in [36] and [47, case s = 1], d = 2 in [49], d = 3 in [50]). Let  $\sigma > 0$ . Up to gauge invariance, the bound states of the NLS with concentrated nonlinearity are of the form

(39) 
$$u_{\beta,\sigma,\omega}^d(x) := q_{\beta,\sigma}^d(\omega) \mathcal{G}_{\omega}^d(x)$$

with

(40) 
$$q_{\beta,\sigma}^{d}(\omega) := \begin{cases} \left(\frac{2^{2\sigma+1}\omega^{\sigma+1/2}}{-\beta}\right)^{1/2\sigma}, & if \quad d=1, \\ \left(\frac{\log(\sqrt{\omega}/2) + \gamma}{-2\pi\beta}\right)^{1/2\sigma}, & if \quad d=2, \\ \left(\frac{\sqrt{\omega}}{-4\pi\beta}\right)^{1/2\sigma}, & if \quad d=3, \end{cases}$$

where

- if d = 1, 3, then  $\omega$  varies in  $(0, +\infty)$  and  $\beta$  varies in  $(-\infty, 0)$ ; while,
- if d = 2, then  $\omega$  varies in  $(0, 4e^{-2\gamma})$  whenever  $\beta$  varies in  $(0, +\infty)$ , and in  $(4e^{-2\gamma}, +\infty)$  whenever  $\beta$  varies in  $(-\infty, 0)$ .

Note that the difference between the odd and the even dimensions is remarkable. Indeed, in d=2 there is a branch of bound states also in the defocusing case  $\beta > 0$ . This is a phenomenon which possesses an analogous neither in the context of the NLS with concentrated nonlinearity nor in the context of the standard NLS.

**Remark 5.1.** In fact, it is also possible to compute explicitly  $E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d)$  and  $M(u_{\beta,\sigma,\omega}^d)$  as functions of the frequency  $\omega$ . As a consequence, one can establish the following qualitative behaviors (that are relevant in the study of the orbital stability discussed below). If d = 1, 3, then

- whenever  $\sigma < 1$ 
  - $-E^d_{\beta,\sigma}(u^d_{\beta,\sigma,\omega})$  is negative, continuous and decreasing on  $\mathbb{R}^+$  and

$$\lim_{\omega \searrow 0} E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d) = 0, \qquad \lim_{\omega \to +\infty} E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d) = -\infty,$$

–  $M(u^d_{\beta,\sigma,\omega})$  is continuous and increasing on  $\mathbb{R}^+$  and

$$\lim_{\omega \searrow 0} M(u_{\beta,\sigma,\omega}^d) = 0, \qquad \lim_{\omega \to +\infty} M(u_{\beta,\sigma,\omega}^d) = +\infty;$$

• whenever  $\sigma > 1$ 

 $-E^d_{\beta,\sigma}(u^d_{\beta,\sigma,\omega})$  is positive, continuous and increasing on  $\mathbb{R}^+$  and

$$\lim_{\omega \searrow 0} E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d) = 0, \qquad \lim_{\omega \to +\infty} E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d) = +\infty,$$

-  $M(u_{\beta,\sigma,\omega}^d)$  is continuous and decreasing on  $\mathbb{R}^+$  and

$$\lim_{\omega \searrow 0} M(u_{\beta,\sigma,\omega}^d) = +\infty, \qquad \lim_{\omega \to +\infty} M(u_{\beta,\sigma,\omega}^d) = 0;$$

- whenever  $\sigma = 1$ ,  $E_{\beta,1}^d(u_{\beta,1,\omega}^d) = 0$  and  $M(u_{\beta,1,\omega}^d) = \mu_{\beta}^d$  (with  $\mu_{\beta}^d$  defined by (38)), for all  $\omega > 0$ . If on the contrary d = 2, then
  - whenever  $\beta < 0$ 
    - $-E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2)$  is continuous on  $(4e^{-2\gamma},+\infty)$ , decreasing on  $(4e^{-2\gamma},4e^{-2\gamma+1/\sigma}]$  and increasing on  $[4e^{-2\gamma+1/\sigma},+\infty)$ , and

$$\lim_{\omega \searrow 4\mathrm{e}^{-2\gamma}} E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2) = 0, \qquad \lim_{\omega \to +\infty} E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2) = +\infty,$$

-  $M(u_{\beta,\sigma,\omega}^2)$  is continuous on  $(4e^{-2\gamma}, +\infty)$ , increasing on  $(4e^{-2\gamma}, 4e^{-2\gamma+1/\sigma}]$  and decreasing on  $[4e^{-2\gamma+1/\sigma}, +\infty)$ , and

$$\lim_{\omega \searrow 4\mathrm{e}^{-2\gamma}} M(u_{\beta,\sigma,\omega}^2) = \lim_{\omega \to +\infty} M(u_{\beta,\sigma,\omega}^2) = 0;$$

- whenever  $\beta > 0$ 
  - $-E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2)$  is continuous and increasing on  $(0,4e^{-2\gamma})$ , and

$$\lim_{\omega \searrow 0} E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2) = -\infty, \qquad \lim_{\omega \nearrow 4e^{-2\gamma}} E_{\beta,\sigma}^2(u_{\beta,\sigma,\omega}^2) = 0,$$

–  $M(u_{\beta,\sigma,\omega}^2)$  is continuous and decreasing on  $(0,4\mathrm{e}^{-2\gamma})$ , and

$$\lim_{\omega \searrow 0} M(u_{\beta,\sigma,\omega}^2) = +\infty, \qquad \lim_{\omega \nearrow 4\mathrm{e}^{-2\gamma}} M(u_{\beta,\sigma,\omega}^2) = 0.$$

### 5.1. Orbital stability

The most natural question concerning the bound states is about their *stability*. In particular, one starts discussing *orbital* stability.

Precisely, a bound state  $u^d_{\beta,\sigma,\omega}$  is said orbitally stable when, for every  $\varepsilon > 0$ , there exists  $\delta > 0$  such that: if  $\|\psi_0 - e^{i\theta}u^d_{\beta,\sigma,\omega}\|_{V_d} < \delta$  for some  $\theta \in [0,2\pi)$  and  $\psi$  is a weak solution of (26) on  $[0,T^d_{\beta,\sigma}(\psi_0))$ , then  $\psi$  can be continued to a solution on  $[0,+\infty)$  and

$$\sup_{t \in \mathbb{R}^+} \inf_{\theta \in [0,2\pi)} \|\psi(t,\cdot) - e^{i\theta} u_{\beta,\sigma,\omega}^d\|_{V_d} < \varepsilon.$$

Otherwise  $u_{\beta,\sigma,\omega}^d$  is said orbitally unstable.

The results on orbital stability established in this context are (mainly) obtained by using the methods introduced by [51] (and the qualitative behaviors pointed our in Remark 5.1). Preliminarily, set

(41) 
$$a_d := \begin{cases} 0, & \text{if } d = 1, 3, \\ 4e^{-2\gamma}, & \text{if } d = 2, \end{cases}$$

**Theorem 5.2** (d = 1 in [36], d = 2 in [49], d = 3 in [50]). Let  $\sigma > 0$  and let  $\beta < 0$ , when d = 1, 3, and  $\beta \in \mathbb{R} \setminus \{0\}$ , when d = 2. Therefore,

- (i) if d=1,3 and  $\sigma<1$ , then  $u^d_{\beta,\sigma,\omega}$  is orbitally stable for every  $\omega>a_d$ ; while, if d=2,  $\beta>0$  and  $\sigma\geqslant 1/2$ , then  $u^2_{\beta,\sigma,\omega}$  is orbitally stable for every  $\omega\in(0,a_2)$ ; (ii) if d=3 and  $\sigma\geqslant 1$  or if d=2,  $\beta<0$  and  $\sigma\geqslant 1/2$ , then  $u^d_{\beta,\sigma,\omega}$  is orbitally unstable for every
- $\omega > a_d$ .

First note that the case  $d=1, \sigma \ge 1$  is not explicitly addressed by the literature. However, this lack is not significant in the sense that it is not due to challenges in the proof, but rather to the fact that historically the discussion on the one-dimensional case mainly focused on other issues. In this case, the result and the proof should be analogous to those of the dimension three.

Note also that, again, the features of the dimension two are remarkable. Here, the transition between stability and instability does not occur at some specific power, i.e. the  $L^2$ -critical power, but switching from defocusing to focusing. As a consequence, here any focusing power nonlinearity can be legitimately considered L<sup>2</sup>-supercritical. On the contrary, the threshold  $\sigma \geq 1/2$  is not significant since, again, it is connected to local well-posedness and is purely technical.

Finally, in strict connection with orbital stability, one can wonder in which cases the bound states are actually ground states at some fixed mass  $\mu > 0$  of the NLS with concentrated nonlinearity, i.e. functions  $u \in V_d$  such that  $||u||_{L^2(\mathbb{R}^d)}^2 = \mu$  and

$$E_{\beta,\sigma}^d(u) = \mathcal{E}_{\beta,\sigma}^d(\mu) := \inf_{\substack{v \in V_d \\ \|v\|_{L^2(\mathbb{R}^d)}^2 = \mu}} E_{\beta,\sigma}^d(v).$$

**Theorem 5.3** (d = 1 in [36], d = 2 in [49]). Let  $\sigma > 0$  and let  $\beta < 0$ , when d = 1, and  $\beta \in \mathbb{R} \setminus \{0\}$ , when d = 2. Therefore,

- (i) if d=1 and  $\sigma<1$  or if d=2 and  $\beta>0$ , then  $\mathcal{E}^d_{\beta,\sigma}(\mu)\in(-\infty,0)$ , for all  $\mu>0$ , and  $u^d_{\beta,\sigma,\omega}$  is the unique (up to gauge invariance) ground state at mass  $M(u^d_{\beta,\sigma,\omega})$  of the NLS with concentrated nonlinearity, for every  $\omega > a_1$  in the former case and every  $\omega \in (0, a_2)$  in the latter case;
- (ii) if d = 1 and  $\sigma = 1$ , then

$$\mathcal{E}_{\beta,1}^{1}(\mu) = \begin{cases} 0, & \text{if} \quad \mu \in (0, \mu_{\beta}^{1}], \\ -\infty, & \text{if} \quad \mu > \mu_{\beta}^{1}, \end{cases}$$

and all the bound states  $(u^1_{\beta,1,\omega})_{\omega>0}$  are actually ground states at mass  $\mu^1_\beta$  of the NLS with concentrated nonlinearity;

(iii) if d = 1 and  $\sigma > 1$  or if d = 2 and  $\beta < 0$ , then  $\mathcal{E}_{\beta,\sigma}^d(\mu) = -\infty$ , for all  $\mu > 0$ .

Combining the results of Theorems 5.1 and 5.3 and of Remark 5.1 one sees that:

- in case (i), there exists a unique positive ground state at mass  $\mu$ , for every  $\mu > 0$ ;
- in case (ii), positive ground states at mass  $\mu$  exist if and only if  $\mu = \mu_{\beta}^{1}$  and are infinitely many;
- in case (iii) no bound state is in fact a ground state.

Finally, we mention that the previous result has never been explicitly discussed for d=3. However, it should be possible to prove a result analogous to the case d=1, using the same techniques.

# 5.2. Asymptotic stability

Another perspective on stability is that of the asymptotic stability, which roughly speaking means that, if the initial datum of (26) is "close" to a bound state, then the solution gets asymptotically "closer and closer" to a, possibly distinct, bound state.

The results mentioned below only deal with the case d=3. Analogous results are obtained for d=1in [52,53], but for another class of nonlinearities, which is very general from a certain point of view, but does not take into account powers. Nevertheless, it is reasonable to guess that one should obtain the same outcomes also for powers. Nothing is known, instead, for d=2.

In order to state the results, some preliminary definitions are required. First, we set  $L_w^1(\mathbb{R}^3) := L^1(\mathbb{R}^3, (1+|x|^{-1}) dx)$ . Moreover, let  $L_\omega^\sigma : L^2(\mathbb{R}^3) \times L^2(\mathbb{R}^3) \to L^2(\mathbb{R}^3) \times L^2(\mathbb{R}^3)$  be the operator defined by

$$L_{\omega}^{\sigma} := \begin{pmatrix} 0 & \mathcal{H}_{\alpha_{1}(\sigma,\omega)}^{3} + \omega \\ -\mathcal{H}_{\alpha_{2}(\sigma,\omega)}^{3} - \omega & 0 \end{pmatrix}$$

where

$$\alpha_1(\sigma,\omega) := -(2\sigma + 1)\frac{\sqrt{\omega}}{4\pi}, \quad \text{and} \quad \alpha_2(\sigma,\omega) := -\frac{\sqrt{\omega}}{4\pi},$$

which is a byproduct of the linearization of  $\mathcal{H}^3_{\beta,\sigma}$  around a bound state of frequency  $\omega$ . Such an operator possesses two purely imaginary eigenvalues  $\pm i\xi_{\sigma,\omega}$ , with  $\xi_{\sigma,\omega} := 2\sigma\omega\sqrt{1-\sigma^2}$  (only defined for  $\sigma \leq 1$ ), whose eigenspaces are spanned by

(42) 
$$\Phi_{\sigma,\omega}(x) := \mathcal{G}^3_{\omega - \xi_{\sigma,\omega}}(x) \begin{pmatrix} 1 \\ i \end{pmatrix} - \frac{\sqrt{1 - \sigma^2} - 1}{\sigma} \mathcal{G}^3_{\omega + \xi_{\sigma,\omega}}(x) \begin{pmatrix} 1 \\ -i \end{pmatrix}$$

and its complex conjugate. Let us also denote by  $\Phi^1_{\sigma,\omega}$  and  $\Phi^2_{\sigma,\omega}$  the first and the second components of  $\Phi_{\sigma,\omega}$ , respectively.

**Theorem 5.4.** Let d = 3,  $\beta < 0$ ,  $\omega > 0$  and  $\theta \in [0, 2\pi)$ . Therefore:

(i) ( [50]) if  $\sigma \in (0, 1/\sqrt{2})$  and  $\psi_0 \in V_3 \cap L^1_w(\mathbb{R}^3)$  with  $\|\psi_0 - e^{i\theta}u^3_{\beta,\sigma,\omega}\|_{V_3 \cap L^1_w(\mathbb{R}^3)}$  sufficiently small, then there exist  $\omega_\infty > 0$ ,  $\psi_\infty : \mathbb{R}^3 \to \mathbb{C}$  and  $r_\infty : \mathbb{R}^+ \times \mathbb{R}^3 \to \mathbb{C}$ , with  $\psi_\infty$ ,  $r_\infty(t) \in L^2(\mathbb{R}^3)$ , for all t > 0, and  $\|r_\infty(t)\|_{L^2(\mathbb{R}^3)} = O(t^{-5/4})$  as  $t \to +\infty$ , such that the weak solution  $\psi$  of (26) satisfies

$$\psi(t,x) = e^{i\omega_{\infty}t} u_{\beta,\sigma,\omega_{\infty}}^{3}(x) + (\mathcal{U}_{3}(t)\psi_{\infty})(x) + r_{\infty}(t,x), \quad as \quad t \to +\infty;$$

(ii) ([54]) there exists  $\sigma^* \in \left(\frac{1}{\sqrt{2}}, \frac{\sqrt{3}+1}{2\sqrt{2}}\right]$  such that, if  $\sigma \in (1/\sqrt{2}, \sigma^*)$  and  $\psi_0 \in V_3 \cap L_w^1(\mathbb{R}^3)$  and satisfies

$$\psi_0 = e^{\imath \omega + \vartheta} u_{\beta,\sigma,\omega}^3 + e^{\imath \omega + \vartheta} \left[ (z + \overline{z}) \Phi_{\sigma,\omega}^1 + \imath (z - \overline{z}) \Phi_{\sigma,\omega}^2 \right] + f$$

for some  $\vartheta \in \mathbb{R}$ ,  $z \in \mathbb{C}$  and  $f \in L^2(\mathbb{R}^3) \cap L^1_w(\mathbb{R}^3)$  such that

$$|z| \leq \sqrt{\varepsilon}$$
, and  $||f||_{L^{1}_{tot}(\mathbb{R}^{3})} \leq C\varepsilon^{3/2}$ 

with C > 0 and  $\varepsilon > 0$  small, then there exist  $\omega_{\infty}$ ,  $k_{\infty} > 0$ ,  $b \in \mathbb{R}$ ,  $\psi_{\infty} : \mathbb{R}^3 \to \mathbb{C}$  and  $r_{\infty} : \mathbb{R}^+ \times \mathbb{R}^3 \to \mathbb{C}$ , with  $\psi_{\infty}$ ,  $r_{\infty}(t) \in L^2(\mathbb{R}^3)$ , for all t > 0, and  $||r_{\infty}(t)||_{L^2(\mathbb{R}^3)} = O(t^{-1/4})$  as  $t \to +\infty$ , such that the weak solution  $\psi$  of (26) satisfies

$$\psi(t,x) = e^{i\omega_{\infty}t + ib\log(1 + \varepsilon k_{\infty}t)}u_{\beta,\sigma,\omega_{\infty}}^{3}(x) + (\mathcal{U}_{3}(t)\psi_{\infty})(x) + r_{\infty}(t,x), \quad as \quad t \to +\infty.$$

#### 6. Blow-up analysis

The last question on (26) discussed by the literature is the so called *blow-up analysis*, that is the search for solutions that *blow-up* in a finite time. By (33), (36) and (37) this is equivalent to prove that  $T_{\beta,\sigma}^d(\psi_0) < +\infty$ .

# 6.1. First results

First blow-up results are based on a method, introduced by [55], which essentially relies on the detection of the properties of the so called *moment of inertia* associated with a solution  $\psi$  of (26) with initial datum  $\psi_0$ , i.e.

$$\mathcal{M}^d(t) := \int_{\mathbb{R}^d} |x \, \psi(t, x)|^2 \, dx.$$

Precisely, in this context the main point is proving that

(43) 
$$\frac{d^2 \mathcal{M}^d(t)}{dt^2} = 8E^d_{\beta,\sigma}(\psi_0) + g^d_{\beta,\sigma}(|q(t)|^2),$$

with

$$g_{\beta,\sigma}^{d}(y) := \begin{cases} 4\beta \frac{\sigma - 1}{\sigma + 1} y^{\sigma + 1}, & \text{if} \quad d = 1, 3, \\ 2\left(\frac{1}{\pi} - \frac{4\beta\sigma y^{\sigma}}{\sigma + 1}\right) y, & \text{if} \quad d = 2. \end{cases}$$

Equality (43) is usually called *Virial Identity* and enables one to establish, in some cases, uniform concavity of  $\mathcal{M}^d$  depending on the energy of the initial datum. Since positivity and uniform concavity are not consistent with globality-in-time it is clear how this method allows to detect the cases in which blow-up solutions arise.

**Remark 6.1.** Note that by Theorem 4.1 blow-up analysis is meaningful only when  $\beta < 0$  and, for d = 1, 3, when  $\sigma \ge 1$ .

Preliminarily, recall that a Schwartz function is a  $C^{\infty}(\mathbb{R}^d)$ -function which decays faster than the reciprocal of any polinomial as  $|x| \to +\infty$  and whose derivatives of any order decay faster than the reciprocal of any polinomial as  $|x| \to +\infty$ .

**Theorem 6.1** (d=1 in [43], d=2 in [56], d=3 in [57]). Let  $\beta<0$ ,  $\sigma>0$  and  $\psi_0\in V_d$  and let  $T^d_{\beta,\sigma}(\psi_0)$  be defined as in (33). Moreover, assume that:

- (i) when  $d = 1, 3, \sigma \geqslant 1$ ; while,
- (ii) when d=2,  $\sigma \geqslant 1/2$  and the regular part of  $\psi_0$  is a Schwartz function.

Then

(44) 
$$E_{\beta,\sigma}^{d}(\psi_{0}) < \inf_{\omega > a_{d}} E_{\beta,\sigma}^{d}(u_{\beta,\sigma,\omega}^{d}) \Longrightarrow T_{\beta,\sigma}^{d}(\psi_{0}) < +\infty,$$

where  $(u_{\beta,\sigma,\omega}^d)_{\omega>a_d}$  is the family of the bound states given by Theorem 5.1 and  $a_d$  is defined by (41).

Some comments are in order. First,  $\inf_{\omega>a_d} E^d_{\beta,\sigma}(u^d_{\beta,\sigma,\omega})$  can be explicitly computed and there results

$$\inf_{\omega > a_d} E_{\beta,\sigma}^d(u_{\beta,\sigma,\omega}^d) = \begin{cases} 0 & \text{if } d = 1, 3, \\ -\frac{\sigma}{4\pi(\sigma+1)(-4\pi\sigma\beta)^{1/\sigma}} & \text{if } d = 2. \end{cases}$$

On the other hand, (44) is sharp in the sense that above this energy threshold one has, for instance, standing waves, which are global-in-time.

Moreover, we underline that in d = 2 the assumptions on  $\sigma$  and on the regularity of the initial datum are just technical. On the contrary, it is remarkable that in this case one can find blowing-up solutions for any focusing power nonlinearity (again, a phenomenon with no analogue neither in the context of the NLS with concentrated nonlinearity nor in the context of the standard NLS).

Finally, note that, when d = 1, 3 and  $\sigma = 1$ , (44) does not contradict point (ii) of Theorem 4.2 since one can actually prove (using some suitable versions of the Gagliardo-Nirenberg inequalities established in [43] for d = 1 and in [45] for d = 3) that in these cases

$$E_{\beta,1}^d(\psi_0) < \inf_{\omega > 0} E_{\beta,1}^d(u_{\beta,1,\omega}^d) = 0 \implies M(\psi_0) > \mu_{\beta}^d,$$

where  $\mu_{\beta}^{d}$  is again the  $L^{2}$ -critical mass defined by (38).

# 6.2. Pseudoconformal invariance

In addition to Theorem 6.1, whenever  $d=1, 3, \beta<0$  and  $\sigma=1$  it is possible to explicitly construct blowing-up solutions whose initial datum satisfies  $M(\psi_0)=\mu_\beta^d$ . This can be done thanks to an additional symmetry arising in the  $L^2$ -critical case, the so called *pseudoconformal invariance*, which can be stated as follows.

**Proposition 6.1** (d = 1 in [58], d = 3 in [57]). Let  $d = 1, 3, \beta \in \mathbb{R} \setminus \{0\}$  and  $\sigma = 1$ . If  $\psi$  is a solution of (26) with initial datum  $\psi_0$ , then, for every fixed T > 0,

$$\Psi_T^d(t,x) := \frac{e^{-i\frac{|x|^2}{4(T-t)}}}{(T-t)^{d/2}}\psi\left(\frac{1}{T-t}, \frac{x}{T-t}\right)$$

is a solution of (26) with initial datum

$$\Psi_{T,0}^d(x) := \frac{e^{-i\frac{|x|^2}{4T}}}{T^{d/2}}\psi\left(\frac{1}{T}, \frac{x}{T}\right).$$

Hence, it is sufficient to apply the previous result with  $\psi$  equal to a standing wave to get blow-up at the  $L^2$ -critical mass.

**Theorem 6.2** (d = 1 in [43,58], d = 3 in [57]). Let  $d = 1, 3, \beta < 0$  and  $\sigma = 1$ . Then, for every fixed  $\omega > 0$ ,  $\theta \in [0, 2\pi)$  and T > 0, the function

$$\Psi^{d}_{T,\beta,\omega}(t,x) := \frac{e^{-i\frac{|x|^2}{4(T-t)} + i\frac{\omega}{T-t} + i\theta}}{(T-t)^{d/2}} u^{d}_{\beta,1,\omega}\left(\frac{x}{T-t}\right),$$

with  $u_{\beta,1,\omega}^d$  a bound state given by Theorem 5.1, is a solution of (26) with initial datum

$$\Psi^d_{T,0,\beta,\omega}(x) := \frac{e^{-i\frac{|x|^2}{4T} + i\frac{\omega}{T} + i\theta}}{T^{d/2}} u^d_{\beta,1,\omega}\left(\frac{x}{T}\right)$$

that blows up at T. In other words  $T^d_{\beta,1}(\Psi^d_{T,0,\beta,\omega}) = T < +\infty$ .

#### 6.3. Further analysis

Concluding the section, we mention some finer results concerning blow-up analysis. Note that such results are available only for d = 1, while everything is open for d = 2, 3. Note also that the following results only concern the focusing case and precisely  $\beta = -1$ , for the sake of simplicity.

The former concerns a lower bound for the blow-up rate of blow-up solutions.

**Theorem 6.3 ( [59]).** Let  $d = -\beta = 1$ ,  $\sigma \ge 1$  and  $\psi_0 \in V_1$  and let  $T^1_{-1,\sigma}(\psi_0)$  be defined as in (33). If  $T^1_{-1,\sigma}(\psi_0) < +\infty$ , then there exists  $C_{\sigma} > 0$  such that

$$\|\partial_x \psi(t,\cdot)\|_{L^2(\mathbb{R})} \geqslant \sqrt{|E(\psi_0)|} \qquad \Longrightarrow \qquad \|\partial_x \psi(t,\cdot)\|_{L^2(\mathbb{R})} \geqslant C_{\sigma} (T^1_{-1,\sigma}(\psi_0) - t)^{-\frac{1-\sigma_c}{2}}$$

with

(45) 
$$\sigma_c := \frac{\sigma + 1}{2\sigma}.$$

Furthermore, it is possible to establish a sharp dichotomy between global existence and blow-up both in the  $L^2$ -supercritical and in the  $L^2$ -critical cases.

**Theorem 6.4** ( [59]). Let  $d = -\beta = 1$ ,  $\sigma \ge 1$  and  $\psi_0 \in V_1$  and let  $T^1_{-1,\sigma}(\psi_0)$  be defined as in (33).

(i) On the one hand, set  $\sigma > 1$ , assume

$$M(\psi_0)^{\frac{1-\sigma_c}{\sigma_c}}E(\psi_0) < M(u_{-1,\sigma,1}^1)^{\frac{1-\sigma_c}{\sigma_c}}E(u_{-1,\sigma,1}^1)$$

and define

$$\eta(t) := \frac{\|\psi_0\|_{L^2(\mathbb{R})}^{\frac{1-\sigma_c}{\sigma_c}} \|\partial_x \psi(t,\cdot)\|_{L^2(\mathbb{R})}}{\|u_{-1,\sigma,1}^1\|_{L^2(\mathbb{R})}^{\frac{1-\sigma_c}{\sigma_c}} \|\partial_x u_{-1,\sigma,1}^1\|_{L^2(\mathbb{R})}}.$$

Therefore:

- if  $\eta(0) < 1$ , then  $T^1_{-1,\sigma}(\psi_0) = +\infty$ ;
- if  $\eta(0) > 1$ , then  $T^1_{-1,\sigma}(\psi_0) < +\infty$  and  $\eta(t) > 1$ , for every  $t \in [0, T^1_{-1,\sigma}(\psi_0))$ .
- (ii) On the other hand, set  $\sigma = 1$ . Therefore:
  - if  $M(\psi_0) < M(u^1_{-1,1,1}) = 2$ , then  $E(\psi_0) > 0$  and

$$\|\partial_x \psi(t,\cdot)\|_{L^2(\mathbb{R})} \le \frac{2E(\psi_0)}{1-M(\psi_0)/2}, \quad \forall t \in [0, T^1_{-1,\sigma}(\psi_0)),$$

so that  $T^1_{-1,\sigma}(\psi_0) = +\infty$ ;

• if  $E(\psi_0) < 0$ , then  $T^1_{-1,\sigma}(\psi_0) < +\infty$ .

**Remark 6.2.** In fact, [59] presents also other results. It establishes some further regularity properties of the weak solutions of (26) and discusses both the phenomenon of mass concentration and the features of blow-up solutions at minimal and near-to-minimal mass in the  $L^2$ -critical case.

It is also possible to establish a *self-similar* structure for blow-up solutions in the  $L^2$ -supercritical case.

**Theorem 6.5** ( [60]). Let  $d = -\beta = 1$ ,  $\sigma > 1$  and let  $T^1_{-1,\sigma}(\psi_0)$  be defined as in (33). Then, the function  $\psi(t,x) := \nu(t)^{1/2\sigma} e^{i\tau(t)} \zeta(\nu(t)x)$ 

weakly solves (26) (for some suitable  $\psi_0 \in V_1$ ) with  $\lim_{t \nearrow T} \nu(t) = +\infty$ , for some T > 0, if and only if there exists h > 0 and  $k \in \mathbb{R}$  such that

$$\nu(t) = \frac{1}{\sqrt{2h(T-t)}}, \qquad \tau(t) = \frac{k}{2h} \log\left(\frac{T}{T-t}\right) + \tau(0)$$

and  $\zeta(z)$  solves

$$(k + ih\sigma_c)\zeta - ih\Lambda_z\zeta - \partial_{zz}^2\zeta - \delta_0|\zeta|^{2\sigma}\eta = 0$$

with  $\sigma_c$  defined by (45).

**Remark 6.3.** Note that [60] also establishes some qualitative properties of the function  $\zeta$ .

Finally, it is worth mentioning a result, which is more connected to *scattering theory* in the cases where global existence is guaranteed, but which is nevertheless strictly related to the blow-up analysis. It concerns the so called *asymptotic completeness*.

**Theorem 6.6 ( [61]).** Let  $d = -\beta = 1$ ,  $\sigma > 1$  and  $\psi_0 \in V_1$ . If

$$M(\psi_0)^{\frac{1-\sigma_c}{\sigma_c}}E(\psi_0) < M(u_{-1,\sigma,1}^1)^{\frac{1-\sigma_c}{\sigma_c}}E(u_{-1,\sigma,1}^1),$$

and

$$\|\psi_0\|_{L^2(\mathbb{R})}^{\frac{1-\sigma_c}{\sigma_c}}\|\partial_x\psi_0\|_{L^2(\mathbb{R})} < \|u_{-1,\sigma,1}^1\|_{L^2(\mathbb{R})}^{\frac{1-\sigma_c}{\sigma_c}}\|\partial_x u_{-1,\sigma,1}^1\|_{L^2(\mathbb{R})}$$

and  $\psi$  is the weak solution of (26) with initial datum  $\psi_0$ , then there exists  $\psi^+ \in V_1$  such that

$$\lim_{t \to +\infty} \|\mathcal{U}_1(-t)\psi(t) - \psi^+\|_{V_1} = 0.$$

**Remark 6.4.** Note that a suitable analogue of Theorem 6.6 holds also in the defocusing case.

# 7. Derivation from the standard NLS equation

As mentioned in the introduction, a rigorous mathematical derivation of the NLS with concentrated nonlinearity from quantum many body systems in presence of impurities is still missing. What is available, at least for d = 1 and, partially, for d = 3, is a derivation from the standard NLS equation with suitably shrinking potentials. For d = 2 nothing is known at the moment.

#### 7.1. The one-dimensional case

In this case the result that can be proven is exactly the expected one: the solution of (26) is the limit of the solution of a standard NLS with a suitably shrinking local potential.

Precisely, consider the following equation

(46) 
$$i\frac{\partial\psi_{\varepsilon}}{\partial t} = -\frac{\partial^{2}\psi_{\varepsilon}}{\partial x^{2}} + \frac{1}{\varepsilon}V\left(\frac{x}{\varepsilon}\right)|\psi_{\varepsilon}|^{2\sigma}\psi_{\varepsilon} \qquad \sigma > 0, \quad \varepsilon > 0, \quad \text{on} \quad \mathbb{R}^{+} \times \mathbb{R}.$$

It can be proved (see [62]) that for any  $\psi_0 \in H^1(\mathbb{R})$  and any  $V \in L^1(\mathbb{R}, (1+|x|dx)) \cap L^{\infty}(\mathbb{R})$ ,

- if  $V \ge 0$  and  $\sigma > 0$ , or
- if V < 0 in at least an open interval and  $\sigma \in (0,2)$  (or  $\sigma = 2$  and  $\|\psi_0\|_{H^1(\mathbb{R})}$  is small),

then the Cauchy problem associated with (46) with initial datum  $\psi_0$  admits a unique global *strong*  $H^1$ -solution, namely (see [63]) a distributional solution in  $C^0_{loc}([0, +\infty); H^1(\mathbb{R}))$  of

(47) 
$$\psi_{\varepsilon}(t,x) = (\mathcal{U}_1(t)\psi_0)(x) - \frac{i}{\varepsilon} \int_0^t \int_{\mathbb{R}} \mathcal{U}_1(t-s,x-y)V\left(\frac{y}{\varepsilon}\right) |\psi_{\varepsilon}(s,y)|^{2\sigma} \psi_{\varepsilon}(s,y) \, dy \, ds,$$

which is the *Duhamel's formulation* of the Cauchy problem associated with (46). Then, we can state the following theorem.

**Theorem 7.1** ( [64]). Let  $\beta \in \mathbb{R} \setminus \{0\}$  and  $V \in L^1(\mathbb{R}, (1+|x|dx)) \cap L^{\infty}(\mathbb{R})$  with  $\beta = \int_{\mathbb{R}} V(x) dx$ . Assume also that  $V \geq 0$  or  $\sigma \in (0,1)$ . Therefore, for every  $\psi_0 \in H^1(\mathbb{R})$ , if  $\psi$  is the weak solution of (26) and  $\psi_{\varepsilon}$  is the strong  $H^1$ -solution of (47), then, for any fixed T > 0,

$$\sup_{t\in[0,T]}\|\psi_{\varepsilon}(t,\cdot)-\psi(t,\cdot)\|_{H^1(\mathbb{R})}\longrightarrow 0, \quad as \quad \varepsilon\searrow 0.$$

Remark 7.1. In fact, in [64] the approximation established by Theorem 7.1 is proved in the case of finitely many nonlinear delta potentials by means of a standard NLS with finitely many shrinking local potentials.

# 7.2. The three-dimensional case

Unfortunately, in the three-dimensional case it is still open whether an analogous of Theorem 7.1 can be proved using local potentials. Nevertheless, it is possible to prove a version of it with a family of nonlocal shrinking potentials.

Precisely, for every fixed  $\sigma$ ,  $\varepsilon > 0$  and  $\beta \in \mathbb{R} \setminus \{0\}$ , consider the following equation

$$(48) \quad i\frac{\partial\psi_{\varepsilon}}{\partial t} = -\Delta\psi_{\varepsilon} + \left(\frac{-\varepsilon}{\ell} + \beta \frac{\varepsilon^{2\sigma+2}}{\ell^{2\sigma+2}} |\langle \rho_{\varepsilon}, \psi_{\varepsilon}(t, \cdot) \rangle_{L^{2}(\mathbb{R}^{3})}|^{2\sigma}\right) \langle \rho_{\varepsilon}, \psi_{\varepsilon}(t, \cdot) \rangle_{L^{2}(\mathbb{R}^{3})} \rho_{\varepsilon}, \quad \text{on} \quad \mathbb{R}^{+} \times \mathbb{R}^{3},$$

where  $(\rho_{\varepsilon})_{\varepsilon>0}$  is defined by

$$\rho_{\varepsilon}(x) := \frac{1}{\varepsilon^3} \rho\left(\frac{x}{\varepsilon}\right),\,$$

with  $\rho$  a real, positive, spherically symmetric Schwartz function of  $\mathbb{R}^3$ , (i.e. a kernel for  $\delta_0$  in  $\mathbb{R}^3$ ) and

$$\ell = \ell(\rho) := \langle \rho, (-\Delta)^{-1} \rho \rangle_{L^2(\mathbb{R}^3)}.$$

In this case, the *Duhamel's formulation* of the Cauchy problem associated with (48) is

$$(49) \quad \psi_{\varepsilon}(t,x) = (\mathcal{U}_{3}(t)\psi_{\varepsilon}(0,\cdot))(x) + \\ -\frac{\imath\varepsilon}{\ell} \int_{0}^{t} \left( -1 + \beta \frac{\varepsilon^{2\sigma+1}}{\ell^{2\sigma+1}} |\langle \rho_{\varepsilon}, \psi_{\varepsilon}(t,\cdot) \rangle_{L^{2}(\mathbb{R}^{3})}|^{2\sigma} \right) \langle \rho_{\varepsilon}, \psi_{\varepsilon}(t,\cdot) \rangle_{L^{2}(\mathbb{R}^{3})} (\mathcal{U}_{3}(t-s)\rho_{\varepsilon})(x) ds.$$

One can prove (see [48]) that, whenever  $\psi(0,\cdot) \in H^2(\mathbb{R}^3)$ , there exists a unique global  $H^2$ -strong solution of (48), namely a solution of (49) in  $C^0_{loc}([0,+\infty);H^2(\mathbb{R}^3)) \cap C^1_{loc}([0,+\infty);L^2(\mathbb{R}^3))$ .

Now, recall that in order to represent  $dom(\mathcal{H}_{\beta,\sigma}^3)$  one can use the nonlinear analogous to (10), i.e.

$$\operatorname{dom}(\mathcal{H}^3_{\beta,\sigma}) = \left\{ u \in L^2(\mathbb{R}^3) : \exists q \in \mathbb{C} \text{ s.t. } u - q\mathcal{G}^3_0 =: \phi \in H^2_{loc}(\mathbb{R}^3) \cap \mathring{H}^1(\mathbb{R}^3) \cap \mathring{H}^2(\mathbb{R}^3), \ \phi(0) = \beta |q|^{2\sigma} q \right\}.$$

Moreover, one can check that ([48])

(50) 
$$\psi_{\varepsilon,0} := \phi_0 + q_0 \rho_{\varepsilon} * \mathcal{G}_0^3 \in H^2(\mathbb{R}^3), \qquad \forall \psi_0 \in \text{dom}(\mathcal{H}_{\beta,\sigma}^3).$$

**Theorem 7.2** ( [48]). Let  $\beta \in \mathbb{R} \setminus \{0\}$  and  $\varepsilon > 0$ . Assume also that  $\sigma > 0$  when  $\beta > 0$ , whereas  $\sigma \in (0,1)$  when  $\beta < 0$ . Therefore, for every  $\psi_0 \in \text{dom}(\mathcal{H}^3_{\beta,\sigma})$ , if  $\psi$  is the weak solution of (26) and  $\psi_{\varepsilon}$  is the  $H^2$ -strong solution of (48) with  $\psi_{\varepsilon}(0,\cdot) = \psi_{\varepsilon,0}$  ( $\psi_{\varepsilon,0}$  being defined by (50)), then, for any fixed T > 0,

$$\sup_{t \in [0,T]} \|\psi_{\varepsilon}(t,\cdot) - \psi(t,\cdot)\|_{L^{2}(\mathbb{R}^{3})} \leqslant C\varepsilon^{\delta},$$

for some constant C > 0 and  $\delta \in (0, 1/4)$ .

# 8. Results on connected problems

Finally, it is worth reporting some works that address problems which are closely related in some sense to the NLS equation with point-concentrated nonlinearities.

# 8.1. Nonlinear perturbations of the fractional Laplacian

First, we mention a "fractional perturbation" in dimension one of what we introduced in Section 2 (nothing has been done in higher dimensions). The unique difference here is that the operator perturbed by a nonlinear Delta potential is the fractional Laplacian  $(-\Delta)^s$ .

The problem is again defined by a suitable modification of the linear model (studied in [65] and partially in [47]), and reads exactly as (26) with  $\mathcal{H}^1_{\beta,\sigma}$  replaced by  $\mathcal{H}^{1,s}_{\beta,\sigma}$ , where  $\mathcal{H}^{1,s}_{\beta,\sigma}$  is an operator whose definition is completely analogous to (27)-(28) with  $\mathcal{G}^1_{\lambda}$  replaced by the Green's function of  $(-\frac{d^2}{dx^2})^s + \lambda$ , denoted by  $\mathcal{G}^{1,s}_{\lambda}$ , and  $-\Delta$  is replaced by  $(-\frac{d^2}{dx^2})^s$ .

**Remark 8.1.** As in the non fractional case, an analogous definition can be constructed from the fractional analogue to (8)-(9)-(10)-(12).

The problem has been addressed by [47], which establishes:

- (i) local well-posedness in strong sense,
- (ii) conservation of mass and energy,
- (iii) global well-posedness in the defocusing and the focusing subcritical case (with critical power here depending on s);
- (iv) existence of blow-up solutions in the focusing supercritical case;
- (v) complete classification of the standing waves.

# 8.2. Nonlinearity concentrated on a sphere

Another variant of the problem introduced by Section 2 arises as one considers a non-zero-dimensional support for the delta measure. For instance, one may think to a smooth, compact manifold without boundary, such as, in particular, the sphere  $\mathbb{S}^2 \subset \mathbb{R}^3$ .

Here, the whole setting is more technical, even in the linear case (see [66] for  $\mathbb{S}^2$  and [67–69] for more complex geometries). However, the proper variant of (26) is obtained by replacing  $\mathcal{H}^3_{\beta,\sigma}$  with the map  $\mathcal{H}^{\mathbb{S}^2}_{\beta,\sigma}$ , with domain

(51) 
$$\operatorname{dom}(\mathcal{H}_{\beta,\sigma}^{\mathbb{S}^2}) := \left\{ u \in L^2(\mathbb{R}^3) : \exists \lambda > 0, \ q : \mathbb{S}^2 \to \mathbb{C} \right. \text{ s.t.}$$

(52) 
$$u + \mathcal{G}_{\lambda}^{3}(\beta|q|^{2\sigma}q) =: \phi_{\lambda} \in H^{2}(\mathbb{R}^{3}) \text{ and } q = u_{|\mathbb{S}^{2}}$$

and action

(53) 
$$\mathcal{H}_{\beta,\sigma}^{\mathbb{S}^2} u = -\Delta \phi_{\lambda} + \lambda \mathcal{G}_{\lambda}^3(\beta |q|^{2\sigma} q), \qquad \forall u \in \text{dom}(\mathcal{H}_{\beta,\sigma}^{\mathbb{S}^2}),$$

where  $\mathscr{G}^3_{\lambda}$  is now the *Green's potential* associated with the unit sphere of the operator  $-\Delta + \lambda$  in  $\mathbb{R}^3$ , i.e.

(54) 
$$\mathscr{G}_{\lambda}^{3}(h)[x] := \int_{\mathbb{S}^{2}} \mathscr{G}_{\lambda}^{3}(x-y) h(y) \ d\mathbb{S}^{2}(y), \qquad \forall h : \mathbb{S}^{2} \to \mathbb{C}, \quad \forall x \in \mathbb{R}^{3}.$$

Remark 8.2. It is possible to find a definition in which the dumb parameter  $\lambda$  does not appear by suitably adapting the sphere versions of (8) and (12). Note that the link with the point-concentrated case in d = 1 is due to the codimension of the support of the delta, which is one in both cases.

The problem has been addressed by [70], which establishes:

- (i) local well-posedness in strong sense,
- (ii) conservation of mass and energy,
- (iii) global well-posedness in the defocusing case for non large exponents of the nonlinearity.

#### 8.3. Non-autonomous point-concentrated delta potentials

Finally, we mention a class of problems where the point delta potentials are linear, but non-autonomous. This class of models is historically mentioned together with nonlinear delta potentials since, even though many features are different, the main techniques used for the investigation are the same.

Such models arise as one sets  $\alpha = \alpha(t)$  in (4)-(5), with  $\alpha : \mathbb{R} \to \mathbb{C}$  a given function, and considers the associated variant of (17). The resulting Cauchy problem involves a family of operators  $(\mathcal{H}^d_{\alpha(t)})_{t\in\mathbb{R}}$ , indexed by the time variable t, which are self-adjoint at any fixed  $t \in \mathbb{R}$ . However, since the operator is not the same at every time, standard theory based on Stone's theorem does not apply any more, thus requiring to rely on some specific techniques used in the context of nonlinear delta potentials.

The main reason of interest for this model descends from an effective description of the microscopic dynamics of a quantum particle interacting with bosonic scalar quantum fields in configurations where the fields are very intense and the average number of carriers is large, in the so called *quasi-classical limit* (see [71,72]). A typical example is given by the description of a quantum particle coupled with both acoustic and optical phonons in a compound ionic crystal (see, e.g., [73, Chapter 4]).

Global well-posedness for this problem has been studied in [74] for d=1, in [75] for d=2 and in [76] for d=3. Anyway, in this context, another interesting issue is the study of the asymptotic behavior of the survival probability of the  $L^2$ -normalized bound state  $\varphi_{\alpha(0)}^d$  associated with the sole eigenvalue of  $\mathcal{H}_{\alpha(0)}^d$  (see after (14)), i.e.

survival probability := 
$$|\langle \varphi_{\alpha(0)}^d, \psi(t, \cdot) \rangle_{L^2(\mathbb{R}^d)}|^2$$

where  $\psi$  is the evolution of the initial state  $\psi_0 = \varphi_{\alpha(0)}^d$ . Such a quantity can be proved to vanish at infinity, which shows a *complete ionization* phenomenon, and an estimate of its decay rate can be also provided ([77,78] for d = 1, [75] for d = 2 and [79] for d = 3).

Finally, we underline that also a different non-autonomous delta model has been managed in the literature: the so-called *traveling deltas* ([80]). In this case, it is no more the strength of the delta potential which depends on time, but the point where the delta measure is based.

# **Fundings**

The author has been partially supported by the INdAM GNAMPA project 2022 "Modelli matematici con singolarità per fenomeni di interazione" (CUP E55F22000270001).

#### References

- 1. F. Dalfovo, S. Giorgini, L. Pitaevskii, and S. Stringari, Theory of Bose-Einstein condensation in trapped gases, *Reviews of Modern Physics*, vol. 71, pp. 463–512, 1999.
- 2. B. Malomed, Nonlinear Schrödinger Equations, in *Encyclopedia of Nonlinear Science* (A. Scott, ed.), pp. 639–643, Routledge, 2005.
- 3. E. Fermi, Sul moto dei neutroni nelle sostanze idrogenate, Ricerca Scientifica, vol. 7, 1936.
- 4. O. Bulashenko, V. Kochelap, and L. Bonilla, Coherent patterns and self-induced diffraction of electrons on a thin nonlinear layer, *Physical Review B*, vol. 54, pp. 1537–1540, 1996.
- 5. G. J. Lasinio, C. Presilla, and J. Sjostrand, On Schrödinger equations with concentrated nonlinearities, *Annals of Physics*, vol. 240, pp. 1–21, 1995.
- 6. B. Malomed and M. Azbel, Modulational instability of a wave scattered by a nonlinear center, *Physical Review B*, vol. 47, pp. 10402–10406, 1993.
- 7. M. Molina and C. Bustamante, The attractive nonlinear delta-function potential, *American Journal of Physics*, vol. 70, pp. 67–70, 2002.
- 8. F.Nier, The dynamics of some quantum open systems with short-range nonlinearities, *Nonlinearity*, vol. 11, pp. 1127–1172, 1998.
- 9. C. Presilla, G. Jona-Lasinio, and F. Capasso, Nonlinear feedback oscillations in resonant tunneling through double barriers, *Physical Review B*, vol. 43, pp. 5200–5203, 1991.
- 10. A. Sukhorukov, Y. Kivshar, and O. Bang, Two-color nonlinear localized photonic modes, *Physical Review E*, vol. 60, pp. R41–R44, 1999.
- 11. A. Sukhorukov, Y. Kivshar, O. Bang, J. Rasmussen, and P. Christiansen, Nonlinearity and disorder: Classification and stability of nonlinear impurity modes, *Physical Review E*, vol. 63, pp. 366011–3660118, 2001.
- 12. P. Yeh, Optical Waves in Layered Media. Wiley, 2005.
- 13. N. Dror and B. Malomed, Solitons supported by localized nonlinearities in periodic media, *Physical Review A*, vol. 83, Paper No. 033828, 2011.
- 14. K. Li, P. Kevrekidis, B. Malomed, and D. Frantzeskakis, Transfer and scattering of wave packets by a nonlinear trap, *Physical Review E*, vol. 84, Paper No. 056609, pp. 1103–1128, 2011.
- 15. H. Sakaguchi and B. Malomed, Singular solitons, *Physical Review E*, vol. 101, Paper No. 012211, 2020.
- 16. E. Shamriz, Z. Chen, B. Malomed, and H. Sakaguchi, Singular mean-field states: A brief review of recent results, *Condensed Matter*, vol. 5, Paper No. 20, 2020.
- 17. R. Adami and D. Noja, Existence of dynamics for a 1D NLS equation perturbed with a generalized point defect, *Journal of Physics. A. Mathematical and Theoretical*, vol. 42, Paper No. 495302, 2009.
- 18. R. Adami and D. Noja, Stability and symmetry-breaking bifurcation for the ground states of a NLS with a  $\delta'$  interaction, Communications in Mathematical Physics, vol. 318, pp. 247–289, 2013.
- 19. R. Adami, D. Noja, and N. Visciglia, Constrained energy minimization and ground states for NLS with point defects, *Discrete and Continuous Dynamical Systems Series B*, vol. 18, pp. 1155–1188, 2013.
- 20. K. Datchev and H. J. Fast soliton scattering by attractive delta impurities, Communications in Partial

- Differential Equations, vol. 34, pp. 1074–1113, 2009.
- 21. R. Fukuizumi and L. Jeanjean, Stability of standing waves for a nonlinear Schrödinger equation with a repulsive Dirac delta potential, *Discrete and Continuous Dynamical Systems*, vol. 21, pp. 121–136, 2008.
- 22. R. Fukuizumi, M. Otha, and T. Ozawa, Nonlinear Schrödinger equation with a point defect, *Annales de l'Institut Henri Poincaré C, Analyse non linéaire*, vol. 25, pp. 837–845, 2008.
- 23. R. Goodman, P. Holmes, and M. Weinstein, Strong NLS soliton-defect interactions, *Physica D: Non-linear Phenomena*, vol. 192, pp. 215–248, 2004.
- 24. J.Holmer, J. Marzuola, and M. Zworski, Fast soliton scattering by delta impurities, *Communications in Mathematical Physics*, vol. 274, pp. 187–216, 2007.
- 25. S. L. Coz, R. Fukuizumi, G. Fibich, B. Ksherim, and Y. Sivan, Instability of bound states of a nonlinear Schrödinger equation with a Dirac potential, *Physica D: Nonlinear Phenomena*, vol. 237, pp. 1103–1128, 2008.
- 26. J. Murphy and K. Nakanishi, Failure of scattering to solitary waves for long-range nonlinear Schrödinger equations, *Discrete and Continuous Dynamical Systems*, vol. 41, pp. 1507–1517, 2021.
- 27. R. Adami, F. Boni, R. Carlone, and L. Tentarelli, Ground states for the planar NLSE with a point defect as minimizers of the constrained energy, *Calculus of Variations and Partial Differential Equations*, vol. 61, Paper No. 195, 2022.
- 28. R. Adami, F. Boni, R. Carlone, and L. Tentarelli, Existence, structure, and robustness of ground states of a NLSE in 3D with a point defect, *Journal of Mathematical Physics*, vol. 63, Paper No. 071501, 2022.
- 29. C. Cacciapuoti, D. Finco, and D. Noja, Well posedness of the nonlinear Schrödinger equation with isolated singularities, *Journal of Differential Equations*, vol. 305, pp. 288–318, 2021.
- 30. C. Cacciapuoti, D. Finco, and D. Noja, Failure of scattering for the NLSE with a point interaction in dimension two and three, arXiv:2212.14216 [math-ph], 2023.
- 31. N. Fukaya, V. Georgiev, and M. Ikeda, On stability and instability of standing waves for 2d-nonlinear Schrödinger equations with point interaction, *Journal of Differential Equations*, vol. 321, pp. 258–295, 2022.
- 32. D. Finco and D. Noja, Blow-up and instability of standing waves for the NLS with a point interaction in dimension two, Zeitschrift für angewandte Mathematik und Physik, vol. 74, Paper No. 162, 2023.
- 33. T. Fülöp and I. Tsutsui, A free particle on a circle with point interaction, *Physics Letters A*, vol. 264, pp. 366–374, 2000.
- 34. F. Boni and R. Carlone, NLS ground states on the half-line with point interactions, *Nonlinear Dif*ferential Equations and Applications NoDEA, vol. 30, Paper No. 51, 2023.
- 35. R. Adami, F. Boni, and S. Dovetta, Competing nonlinearities in NLS equations as source of threshold phenomena on star graphs, *Journal of Functional Analysis*, vol. 283, Paper No. 109483, 2022.
- 36. F. Boni and S. Dovetta, Prescribed mass ground states for a doubly nonlinear Schrödinger equation in dimension one, *Journal of Mathematical Analysis and Applications*, vol. 496, Paper No. 124797, 2021
- 37. F. Boni and S. Dovetta, Doubly nonlinear Schrödinger ground states on metric graphs, *Nonlinearity*, vol. 35, pp. 3283–3323, 2022.
- 38. S. Albeverio, F. Gesztesy, R. Høegh-Krohn, and H. Holden, *Solvable Models in Quantum Mechanics*. Springer-Verlag, 1988.
- 39. R. Adami, A Class of Schrödinger Equations with Concentrated Nonlinearity. PhD thesis, Department of Mathematics "Guido Castelnuovo", 2000.
- 40. M. Abramovitz and I. Stegun, Handbook of Mathematical Functions: with Formulas, Graphs, and Mathematical Tables. Dover, 1965.
- 41. M. Reed and B. Simon, Methods of modern mathematical physics. II. Fourier analysis, self-adjointness. Academic Press, 1975.

- 42. S. Albeverio, Z. Brzežniak, and L. Dabrowski, Time-dependent propagator with point interaction, Journal of Physics. A. Mathematical and General, vol. 27, pp. 4933–4943, 1994.
- 43. R. Adami and A. Teta, A class of nonlinear Schrödinger equations with concentrated nonlinearity, *Journal of Functional Analysis*, vol. 180, pp. 148–175, 2001.
- 44. C. Carlone, M. Correggi, and L. Tentarelli, Well-posedness of the two-dimensional nonlinear Schrödinger equation with concentrated nonlinearity, *Annales de l'Institut Henri Poincaré C, Analyse non linéaire*, vol. 36, pp. 257–294, 2019.
- 45. R. Adami, G. Dell'Antonio, R. Figari, and A. Teta, The Cauchy problem for the Schrödinger equation in dimension three with concentrated nonlinearity, *Annales de l'Institut Henri Poincaré C, Analyse non linéaire*, vol. 20, pp. 477–500, 2003.
- 46. C. Carlone, A. Fiorenza, and L. Tentarelli, The action of Volterra integral operators with highly singular kernels on Hölder continuous, Lebesgue and Sobolev functions, *Journal of Functional Analysis*, vol. 273, pp. 1258–1294, 2017.
- 47. C. Carlone, D. Finco, and L. Tentarelli, Nonlinear singular perturbations of the fractional Schrödinger equation in dimension one, *Nonlinearity*, vol. 32, pp. 3112–3143, 2019.
- 48. C. Cacciapuoti, D. Finco, D. Noja, and A. Teta, The point-like limit for a NLS equation with concentrated nonlinearity in dimension three, *Journal of Functional Analysis*, vol. 273, pp. 1762–1809, 2017.
- 49. R. Adami, R. Carlone, M. Correggi, and L. Tentarelli, Stability of the standing waves of the concentrated NLSE in dimension two, *Mathematics in Engineering*, vol. 3, pp. 1–15, 2021.
- 50. R. Adami, D. Noja, and C. Ortoleva, Orbital and asymptotic stability for standing waves of a nonlinear Schrödinger equation with concentrated nonlinearity in dimension three, *Journal of Mathematical Physics*, vol. 54, Paper No. 013501, 2013.
- 51. M. Grillakis, J. Shatah, and W. Strauss, Stability theory of solitary waves in the presence of symmetry. I, *Journal of Functional Analysis*, vol. 74, pp. 160–197, 1987.
- 52. V. Buslaev, A. Komech, E. Kopylova, and D. Stuart, On asymptotic stability of solitary waves in Schrödinger equation coupled to nonlinear oscillator, *Communications in Partial Differential Equations*, vol. 33, pp. 669–705, 2008.
- 53. A. Komech, E. Kopylova, and D. Stuart, On asymptotic stability of solitons in a nonlinear Schrödinger equation, *Communications on Pure and Applied Analysis*, vol. 11, pp. 1063–1079, 2012.
- 54. R. Adami, D. Noja, and C. Ortoleva, Asymptotic stability for standing waves of a NLS equation with subcritical concentrated nonlinearity in dimension three: neutral modes, *Discrete and Continuous Dynamical Systems Series B*, vol. 36, pp. 5837–5879, 2016.
- 55. R. Glassey, On the blowing up of solutions to the Cauchy problem for nonlinear Schrödinger equations, *Journal of Mathematical Physics*, vol. 18, pp. 1794–179, 1977.
- 56. R. Adami, R. Carlone, M. Correggi, and L. Tentarelli, Blow-up for the pointwise NLS in dimension two: absence of critical power, *Journal of Differential Equations*, vol. 269, pp. 1–37, 2020.
- 57. R. Adami, G. Dell'Antonio, R. Figari, and A. Teta, Blow-up solutions for the Schrödinger equation in dimension three with a concentrated nonlinearity, *Annales de l'Institut Henri Poincaré C, Analyse non linéaire*, vol. 21, pp. 121–137, 2004.
- 58. R. Adami and A. Teta, A simple model of concentrated nonlinearity, *Operator Theory: Advances and Applications*, vol. 108, pp. 183–189, 1999.
- 59. J.Holmer and C.Liu, Blow-up for the 1d nonlinear Schrödinger equation with point nonlinearity I: Basic theory, *Journal of Mathematical Analysis and Applications*, vol. 483, Paper No. 123522, 2020.
- 60. J.Holmer and C.Liu, Blow-up for the 1d nonlinear Schrödinger equation with point nonlinearity II: supercritical blow-up profiles, *Communications on Pure and Applied Analysis*, vol. 20, pp. 215–242, 2021.
- 61. R. Adami, R. Fukuizumi, and J. Holmer, Scattering for the  $L^2$ -supercritical point NLS, Transactions of the American Mathematical Society, vol. 374, pp. 35–60, 2021.
- 62. T. Cazenave, Semilinear Schrödinger equations. American Mathematical Society, 2003.

- 63. T. Tao, Nonlinear dispersive equations. Local and global analysis. American Mathematical Society, 2006.
- 64. C. Cacciapuoti, D. Finco, D. Noja, and A. Teta, The NLS equation in dimension one with spatially concentrated nonlinearities: the pointlike limit, *Letters in Mathematical Physics*, vol. 104, pp. 1557–1570, 2014.
- 65. A. Michelangeli, A. Ottolini, and R. Scandone, Fractional powers and singular perturbations of quantum differential Hamiltonians, *Journal of Mathematical Physics*, vol. 59, Paper No. 072106, 2018.
- 66. J. Antoine, F. Gesztesy, and J. Shabani, Exactly solvable models of sphere interactions in quantum mechanics, *Journal of Physics. A. Mathematical and General*, vol. 20, pp. 3687–3712, 1987.
- 67. J. Behrndt, P. Exner, and V. Lotoreichik, Schrödinger operators with δ-interactions supported on conical surfaces, Journal of Physics. A. Mathematical and Theoretical, vol. 47, Paper No. 355202, 2014.
- 68. J. Behrndt, P. Exner, M. Holzmann, and V. Lotoreichik, Approximation of Schrödinger operators with  $\delta$ -interactions supported on hypersurfaces, *Mathematische Nachrichten*, vol. 290, pp. 1215–1248, 2017.
- 69. J. Behrndt, M. Langer, and V. Lotoreichik, Schrödinger operators with  $\delta$  and  $\delta'$ -potentials supported on hypersurfaces, *Annales Henri Poincaré*, vol. 14, pp. 385–423, 2013.
- 70. D. Finco, L. Tentarelli, and A. Teta, Well-posedness of the three-dimensional NLS equation with sphere-concentrated nonlinearity, arXiv:2209.13504 [math.AP], 2022.
- 71. C. Carlone, M. Correggi, M. Falconi, and M. Olivieri, Emergence of time-dependent point interactions in polaron models, SIAM Journal on Mathematical Analysis, vol. 53, pp. 4657–4691, 2021.
- 72. M. Correggi, M. Falconi, and M. Olivieri, Quasi-classical dynamics, *Journal fo the European Mathematical Society*, vol. 25, pp. 731–783, 2023.
- 73. C. Kittel, Introduction to Solid State Physics 8th edn. Wiley, 2004.
- 74. T. Hmidi, A. Mantine, and F. Nier, Time-dependent delta-interactions for 1d Schrödinger Hamiltonians, *Mathematical Physics*, *Analysis and Geometry*, vol. 13, pp. 83–103, 2010.
- 75. W. Borrelli, R. Carlone, and L. Tentarelli, Complete ionization for a non-autonomous point interaction model in d=2, Communications in Mathematical Physics, vol. 395, pp. 963–1005, 2022.
- M. Sayapova and D. Yafaev, The evolution operator for time-dependent potentials of zero radius, Akademiya Nauk SSSR. Trudy Matematicheskogo Instituta imeni V. A. Steklova, vol. 159, pp. 167– 174, 1983.
- 77. O. Costin, R. Costin, J. Lebowitz, and A. Rokhlenko, Evolution of a model quantum system under time periodic forcing: conditions for complete ionization, *Communications in Mathematical Physics*, vol. 221, pp. 1–26, 2001.
- 78. O. Costin, R. Costin, and J. Lebowitz, Nonperturbative time dependent solution of a simple ionization model, *Communications in Mathematical Physics*, vol. 361, pp. 217–238, 2018.
- 79. M. Correggi, G. Dell'Antonio, R. Figari, and A. Mantile, Ionization for three dimensional time-dependent point interactions, *Communications in Mathematical Physics*, vol. 257, pp. 169–192, 2005.
- 80. G. Dell'Antonio, R. Figati, and A. Teta, The Schrödinger equation with moving point interactions in three dimensions, in *Stochastic processes*, *physics and geometry: new interplays*, *I* (F. Gesztesy, H. Holden, J. Jost, S. Paycha, M. Röckner, and S. Scarlatti, eds.), pp. 99–113, American Mathematical Society, 2000.